Published in final edited form as: *Remote Sens Environ.* 2020 June ; 242: 111758. doi:10.1016/j.rse.2020.111758.

Crop nitrogen monitoring: Recent progress and principal developments in the context of imaging spectroscopy missions

Katja Berger^{a,*}, Jochem Verrelst^b, Jean-Baptiste Féret^c, Zhihui Wang^d, Matthias Wocher^a, Markus Strathmann^a, Martin Danner^a, Wolfram Mauser^a, Tobias Hank^a

^aDepartment of Geography, Ludwig-Maximilians-Universitaet München, Luisenstr 37, 80333 Munich, Germany

^bImage Processing Laboratory (IPL), Parc Científic, Universitat de València, Paterna, València 46980, Spain

CTETIS, INRAE, AgroParisTech, CIRAD, CNRS, Université Montpellier, Montpellier, France

^dDepartment of Forest and Wildlife Ecology, University of Wisconsin-Madison, 1630 Linden Drive, Madison, WI 53706, USA

Abstract

Nitrogen (N) is considered as one of the most important plant macronutrients and proper management of N therefore is a pre-requisite for modern agriculture. Continuous satellite-based monitoring of this key plant trait would help to understand individual crop N use efficiency and thus would enable site-specific N management. Since hyperspectral imaging sensors could provide detailed measurements of spectral signatures corresponding to the optical activity of chemical constituents, they have a theoretical advantage over multi-spectral sensing for the detection of crop N. The current study aims to provide a state-of-the-art overview of crop N retrieval methods from hyperspectral data in the agricultural sector and in the context of future satellite imaging spectroscopy missions. Over 400 studies were reviewed for this purpose, identifying those estimating mass-based N (N concentration, N%) and area-based N (N content, Narea) using hyperspectral remote sensing data. Retrieval methods of the 125 studies selected in this review can be grouped into: (1) parametric regression methods, (2) linear nonparametric regression methods or chemometrics, (3) nonlinear nonparametric regression methods or machine learning regression algorithms, (4) physically-based or radiative transfer models (RTM), (5) use of alternative data sources (sun-induced fluorescence, SIF) and (6) hybrid or combined techniques. Whereas in the last decades methods for estimation of Narea and N% from hyperspectral data have been mainly based on simple parametric regression algorithms, such as narrowband vegetation indices, there is an increasing trend of using machine learning, RTM and hybrid techniques. Within plants, N is invested in proteins and chlorophylls stored in the leaf cells, with the proteins being the major nitrogen-containing biochemical constituent. However, in most studies, the relationship between

^{*}Corresponding author. katja.berger@lmu.de (K. Berger). Edited by Jing M. Chen

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

N and chlorophyll content was used to estimate crop N, focusing on the visible-near infrared (VNIR) spectral domains, and thus neglecting protein-related N and reallocation of nitrogen to non-photosynthetic compartments. Therefore, we recommend exploiting the estimation of nitrogen via the proxy of proteins using hyperspectral data and in particular the short-wave infrared (SWIR) spectral domain. We further strongly encourage a standardization of nitrogen terminology, distinguishing between N% and N_{area}. Moreover, the exploitation of physically-based approaches is highly recommended combined with machine learning regression algorithms, which represents

an interesting perspective for future research in view of new spaceborne imaging spectroscopy sensors.

Keywords

Hyperspectral; Biochemical traits; Radiative transfer modelling; Hybrid techniques; machine learning

1 Introduction

The determination of nitrogen (N) in agriculture has been discussed and studied intensely because N is one of the most important plant macronutrients strongly influencing crop growth, production and quality (Baret et al., 2007; Lemaire et al., 2008). The proper management of N is a pre-requisite for modern agriculture: optimal crop yield from high quality grain can only be obtained with sufficient provision and corresponding uptake of N. Thus, in agricultural practise, N is usually applied via fertilization (Beeckman et al., 2018). Because the actual uptake of N largely depends on the current growth conditions, which are extremely variable in space and time, fertilization is often oversupplied (Skiba and Rees, 2014). Such excessive fertilization causes plant stress and overproduction of leaves, which in turn renders the plants susceptible to diseases (Powell and Lindquist, 1997). Overfertilization also leads to excess N that remains unused in the soil. If this N leaches below the root zone or is lost through run-off, nitrate (NO_3^{-}) will accumulate in natural water bodies (Jaynes et al., 2001; Padilla et al., 2018). The ecological effects of nutrient overloads are well known as algal blooms or eutrophication of freshwater lakes and coastal areas. This further leads to nitrate contaminated drinking water, which requires expensive treatments. Additionally, nitrous oxide (N₂O) emissions from denitrification and manure decomposition processes on agricultural sites add to the greenhouse gas budget of the atmosphere contributing to global warming (Skiba and Rees, 2014). In summary, both wet deposited forms of N, i.e. ammonium (NH4⁺) and NO3⁻, and gaseous ammonia are harmful to vegetation and therefore have a strong impact on biodiversity through foliar damage, eutrophication, acidification and other stresses. These problems go far beyond agricultural areas: sensitive species are affected within vulnerable habitats often adapted to low nutrient levels, such as grassland, heathland, peatland, forests, arctic and montane ecosystems (Dise et al., 2011).

Nitrogen deficiency, on the other hand, negatively affects photosynthetic assimilation and crop yield both in terms of quantity and quality (Chlingaryan et al., 2018; Jay et al., 2017; Milford et al., 1985). Therefore, fertilizer application rates for optimal economic

and environmental yield should be considered by taking into account the individual needs of genotypes as well as their actual uptake rates depending on growth stage, soil and weather conditions. Moreover, instead of applying excessive amounts of fertilizer to cover potential crop N demand, the N required according to the target yield should be determined (Lemaire et al., 2008). While monitoring crop N content during the early vegetative growth stages is of major importance for the planning of fertilization measures, assessment of N during mature growth stages provides valuable indication of expected yield quality. Thus, a continuous monitoring of crop N during all growth stages could be of high economic impact (Hank et al., 2019).

With this background in mind, our review study has the following objectives:

- We intend to give an actual overview of crop nitrogen retrieval by means of hyperspectral data and related methods both in view of upcoming spaceborne imaging spectroscopy missions and agricultural monitoring.
- A detailed analysis of the literature will indicate common methods used in the past and identify possible gaps and future trends.
- Finally, the review aims to give recommendations and inspirations for possible new research directions going beyond the traditional methods, and to discuss perspectives of N retrieval from future imaging spectroscopy data streams with associated challenges.

The remainder of the paper is organized as follows. Section 2 introduces the physiology of N and the strong relation to proteins. Section 3 presents the exploitation of reflectance data for the estimation of N or proteins and discusses important N related issues encountered in remote sensing based precision farming. The section also summarizes findings from other review studies. Section 4 discusses the current state of remote sensing retrieval algorithms linked to the estimation of N; Section 5 considers future available satellite missions suitable for N monitoring; Section 6 explains the screening of N studies, classifies applied methods and discusses advantages and limitations of the diverse approaches. Finally, Section 7 summarizes significant research gaps and highlights future perspectives.

2 Nitrogen physiology: constituent of chlorophylls and proteins

The biochemical components of leaves are mainly composed of four elements: hydrogen (H), carbon (C), oxygen (O) and nitrogen (N) with resulting optical properties due to interactions of light with C—O, O—H, C—H, and N—H bonds and stretches (Kokaly and Clark, 1999). N, being taken up by the roots from soil in the form of NH_4^+ and NO_3^- , is a rather small component of leaf dry weight, ranging from 0.3% to 6.4% (Wright et al., 2004). A large amount of N is invested in proteins and chlorophylls within the leaf cells, with the proteins being the major N-containing biochemical constituent of plants (Kokaly et al., 2009). Ribulose-1,5-bisphosphate carboxylase/oxygenase (rubisco) is the most abundant protein in leaves and in the world (Ellis, 1979). It catalyses the carbon fixation at the beginning of the Calvin-Benson cycle (Bassham et al., 1954) and accounts for 15–30% of the total leaf N in C3 species (Evans, 1989; Makino, 2003). Moreover, rubisco is a major source for N remobilization along with other photosynthesis-related proteins

(Masclaux-Daubresse et al., 2010). Chlorophylls are the primary light harvesting molecules in the photosynthesis process converting carbon dioxide and water into carbohydrates (Evans, 1989; Kokaly et al., 2009). The organic molecule of chlorophyll is a porphyrin that contains four N atoms, which act as a stabilizer of the central magnesium ion. However, only very small shares of leaf N (1.7%) are held in chlorophyll pigments (Kokaly et al., 2009). In total, about 19% of leaf N in C3 plants is allocated to light harvesting complexes (Chapin et al., 1987), see also Fig. 1. They are mainly used to synthesise chlorophyll and chlorophyll-binding proteins in both photosystems and thus explain the strong correlation between N and chlorophyll proposed by many studies. The largest part of leaf N (70%) is bound in molecules supporting carbon fixation, which include biosynthetic (~18%) and CO_2 -fixing molecules such as rubisco (26%), bioenergetics (5%) and light harvesting (19%), see also Fig. 1 for an exemplarily sun leaf of a C3 plant (Chapin et al., 1987; Kokaly et al., 2009).

Vegetation growth is not a static but a dynamic process of constant nitrogen turnover (Kattge, 2002). For example, early in a growing season N is first bound in vegetative tissues. During the reproductive phase, N is moved or reallocated from the vegetative organs (leaves) to reproductive structures, such as seeds, ears or fruits (Ohyama, 2010). Note that allocation is not an independent process, but the combination of multiple processes involving the assembly and disassembly of N-containing structures, such as proteins, nucleic acids, chlorophyll and phytochemicals, and particularly alkaloids.

Numerous studies have related different kinds of spectroscopic estimation of chlorophyll content to nitrogen, relying on the fact that the two traits are highly correlated. However, the linkage is weak with Pearson correlation coefficient of 0.65 ± 0.15 across ecosystems (Homolová et al., 2013), which can be explained by the minor contribution of N to light harvesting processes, and the different allocations of N, including photosynthesis, storage, respiration and structure (Xu et al., 2012). Some ecosystems, e.g. as the humid tropics, are known for strong decoupling of leaf chlorophyll and N contents (Asner and Martin, 2009). The use of chlorophyll content as proxy for N can be misleading when in reality other mineral deficiencies occur: for instance, the sulphur deficiency symptom leads to chlorosis starting in the emergence of young leaves (Schnug and Haneklaus, 2005). Rust infestation may also be confused with N deficiency, as diseased plants exhibit higher reflectance values in the visible region caused by lower chlorophyll activity (Bravo et al., 2003). Yellowing due to a decrease in chlorophyll of crop leaves may therefore be caused by many factors that are not necessarily related to N deficiency, for instance herbicide injury, liquid fertilizer burn, freeze injury or moisture stress (Kimura et al., 2016). Therefore, decoupling the estimation of chlorophyll and N content from remote sensing data may contribute to improve our capacity to discriminate among various stresses and pathogens. In contrast to chlorophyll, the leaf protein content appears as a strong proxy for leaf N content which may not suffer from the same downsides, as the largest part of leaf N is bound in proteins. Proteins contribute to the different allocations of N (Xu et al., 2012), and strong correlations between N resorption and protein degradation have been evidenced for some species (Yasumura et al., 2007), even if this mechanism is not universally shared among species (Sample and Babst, 2018). The measurement of protein content in food material (including plant tissues) is usually directly derived from the Kjeldahl method which consists in digesting proteins

and measuring the N contained in these proteins, as well as ammonia and ammonium. Traditionally, a nitrogen-to-protein conversion factor of 6.25 was used. Yet, some studies have demonstrated that the factor is invalid for plant materials due to the existence of non-protein nitrogenous compounds (Milton and Dintzis, 1981; Mosse, 1990; Yeoh and Wee, 1994). In this context, different factors have been discussed as reviewed by Mariotti et al. (2008) and a factor of 4.43 was proposed by Yeoh and Wee (1994) for plant tissues after analysis of 90 plant species.

An exemplary relationship between in situ measured canopy chlorophyll content and aboveground nitrogen content (N_{area}) depending on the growth stage of winter wheat and maize is presented in Fig. 2. This finding points toward the limitation of solely using empirical chlorophyll-N relationships to estimate the area-based N content of crop canopies: the correlation between canopy chlorophyll content and aboveground N_{area} decreases as soon as flowers and fruits appear or senescence emerges. This results from the dynamic process of nitrogen turnover, explained before: in the reproductive growth stage, N is reallocated from the vegetative organs to reproductive structures (Ohyama, 2010). Hence, the amount of N in the plant remains the same whereas leaf chlorophyll content decreases.

The relationship among leaf constituents of the Leaf Optical Properties Experiment, LOPEX dataset (Hosgood et al., 1994), confirms these findings: the analysis of the LOPEX data (downloaded from opticleaf.ipgp.fr) revealed a weak correlation (R^2 = 0.14) between N and chlorophylls across growth stages and species, which limits the capability of estimating N from chlorophylls. Hereby, in situ observations for > 40 species were collected, including agriculturally relevant crops such as corn, sunflower, rice, potato, alfalfa and cabbage. This illustrates the difficulty to produce robust models for the estimation of plant N based solely on the relation between chlorophyll and nitrogen.

To summarize the physiological discussion of plant-bound nitrogen, we recommend recognizing proteins as the major N-containing biochemical constituent of leaves and plants. The often assumed strong positive relationship between N and chlorophylls is hampered by the fact that only a small part of the total plant N amount is effectively bound in chlorophylls. Moreover, the multiple allocations and remobilization of N leads to a decrease in the relation between chlorophyll content and N under environmental constraints and with beginning senescence at the end of the growing season. Besides, a positive N–chlorophyll relationship is species-specific and thus not applicable for regions with high species diversity (Hallik et al., 2009; Homolová et al., 2013).

3 Hyperspectral sensing and applications in the context of nitrogen

The three processes of absorption, reflection and transmission describe interactions between incident radiation, leaf biochemical constituents and canopy biophysical traits. The reflected solar radiation in the optical domain (i.e. from 380 to 2500 nm) is generally used in studies that analyze vegetation characteristics, since most of the absorption features of green vegetation are located in this part of the spectrum (Homolová et al., 2013; Kokaly et al., 2009). The reflectance in the optical domain can be divided into three spectral parts: the visible (380–700 nm, VIS), which is dominated by absorption of foliar photosynthetic

pigments, such as chlorophylls, carotenoids and anthocyanins; the near infrared (700–1300 nm, NIR), which is dominated by scattering occurring at both leaf and canopy scale and driven, among others, by leaf structure, the leaf area index (LAI) and plant density; the shortwave infrared region (1300–2500 nm, SWIR) which is dominated by absorption of water, lignin, cellulose and proteins. The studies of Thenkabail et al. (2013a) and Hank et al. (2019) provide illustrative figures of the spectral domains and their sensitivity to specific vegetation variables.

For more than three decades, reflectance data from hyperspectral sensors have been used to estimate and monitor biochemical constituents and biophysical traits of vegetation, including foliar N (Hansen and Schjoerring, 2003; Inoue et al., 2012; Peñuelas et al., 1994). The term hyperspectral, used here for simplicity, rather emphasizes a large number of spectral bands without further specification. The more appropriate term would be "imaging spectrometry" or "imaging spectroscopy" according to Schaepman et al. (2006). It can be understood as a passive remote sensing technology simultaneously acquiring spatially co-registered images or signatures in contiguous series of narrow wavebands. The narrow bands are usually provided in spectral sampling distance (SSD) of 10 nm or less. The availability of hyperspectral data does not always imply that the full spectrum was exploited. Instead, one must distinguish between those studies that used data from hyper-spectral sensors for broadband analysis, i.e. resampling of spectral reflectance according to the spectral configuration of different (broad) band sensor systems, e.g. by Prey and Schmidhalter (2019). Hyper-spectral narrowband remote sensing techniques aim at taking advantage of hundreds of bands sensitive to biochemical constituents and biophysical traits (Marshall et al., 2016). There have been numerous studies demonstrating the suitability of broadband methods (and sensors) for the retrieval of structural biophysical vegetation traits, such as LAI (Darvishzadeh et al., 2019; Richter et al., 2012b; Vuolo et al., 2010). However, the usage of broadband methods (and thus sensors) is limited for the estimation of subtle features related to vegetation biochemistry such as proteins due to the fact that only portions of the solar spectrum are exploited and that spectral information is lost at integrating over coarse bandwidths (Ollinger, 2010). For the detection of these fine features, narrowband analysis or chemometric methods (Lavine and Workman, 2013) should be given preference.

In the pioneering work from Curran (1989), distinct absorption peaks of proteins (and nitrogen) were reported, located mainly in the SWIR spectral range (Table 1). These results were confirmed by Fourty et al. (1996), who found a good agreement within a 10 nm - window of protein absorption positions between their own measurements and the bands identified by Curran (1989). Table 1 summarizes specific absorption bands associated with proteins and nitrogen according to Kumar et al. (2001), Curran (1989) and Fourty et al. (1996). The studies reported absorption bands published by previous studies measuring biochemical absorption features (Curran, 1989; Elvidge, 1990; Himmelsbach et al., 1988; Williams and Norris, 1987).

Some of the indicated absorption bands are not only sensitive to proteins or N, but also to other important biochemical components, such as lignin at 1690 nm and 1940 nm, cellulose at 1940 nm and starch at 1690 nm and 1940 nm. In particular, the strong O—H bound at 1940 nm causes absorption of these biochemicals and water simultaneously (Kumar et

al., 2001). This should be taken into account for the estimation of foliar biochemistry from single absorption features.

Biophysical chemical analyses of N are costly, laborious, time consuming and restricted to a limited number of locations and time points, which does not allow for monitoring continuous seasonal development. Instead, this can be accomplished by usage of precision farming techniques. Precision farming is an important component of smart farming, which aims at information-driven optimization of farming systems (Bach et al., 2016) and includes real-time Earth observation contributing to site-specific planning of important management measures such as fertilization (Cidad et al., 2000; Hank et al., 2019). Recently, such satellite-, planes-, drones or vehicle-based remote sensing technologies are becoming important tools in supporting the present management systems. From satellites, multi-spectral imaging systems have been frequently used to obtain information about crop N status (Söderström et al., 2017; Zheng et al., 2018). Mulla (2013) identified the use of hyperspectral remote sensing for variable rate, inseason management of nitrogen fertilizer as "perhaps of greatest interest for precision agriculture". The implementation of such technologies may also mitigate the negative effects of N on the plant.

Review studies of N retrieval discussing the plethora of multi-spectral systems have been published by Goffart et al. (2008) or Muñoz-Huerta et al. (2013). Both studies highlight the potential of future hyperspectral sensors for N deficiency detection since discrimination of leaf biochemical components can best be achieved by exploiting narrow spectral bands. Moreover, the study of Baret et al. (2007) reviewed the quantification of stress levels from remote sensing observations with N as a key factor. The authors mainly focused on canopy chlorophyll content as a proxy for N based on findings of their previous study (Baret and Fourty, 1997), stating that "there were very few chances to retrieve canopy nitrogen directly or protein content from remote sensing observations, even using hyper-spectral systems" (Baret et al., 2007). This is in accordance with the exhaustive review study of Homolová et al. (2013) who examined optical based plant traits mapping including N. The authors discussed the problem of different scales, such as scaling up from leaf to canopy level and agreement between different kinds of remote sensing observations. Moreover, they presented a comprehensive figure of spectral wavelengths frequently used in scientific literature for estimation of N concentration (N%) and content (Narea), showing a broad distribution over the entire optical domain with some pronounced features in the SWIR. As conclusion of their study, Homolová et al. (2013) strongly supported the hypothesis that optical remote sensing of chlorophyll content can be used as an operational proxy for N estimation, based on moderate to good relationships between nitrogen and chlorophyll. In fact, since decades, numerous studies have been published demonstrating a success in N retrieval using chlorophyll as proxy, e.g. Filella et al. (1995). Dale et al. (2013) reviewed various applications of near-infrared hyperspectral imaging (NIR-HSI) in agriculture with plant stresses imposed, among others, by insufficient N fertilization. The authors pointed out that hyperspectral data from macroscopic to satellite level are currently a large opportunity to assess the quality of agro-food products. Weiss et al. (2020) defined N as a secondary agronomic state variable not directly derivable from radiative transfer modelling. Instead, N results from the combination of several processes within the soil-plant-atmosphere continuum and can only be directly derived using empirical approaches. Deterministic

approaches can be used to estimate primary variables (such as chlorophyll content or proteins) by combining land surface process models and remote sensing data. Knyazikhin et al. (2013b) analyzed the link between vegetation canopy bidirectional reflectance factor (BRF) and foliar N% for forests. The authors concluded that three-dimensional (3-D) radiative transfer theory is required to explain the physically consistent linkage between leaf scattering and canopy reflectance, which often neglected by existing approaches. Multiple studies have demonstrated that N% of grasslands or cultivates decreases over the growing season, implying that it negatively relates to dry total biomass (Greenwood et al., 1986; Justes et al., 1994; Tilly and Bareth, 2019). This dropping of crop-specific critical N% during growth can be described by a negative power function, named "dilution curve". See also Lemaire et al. (2008) for details. In this context, the concept of the nitrogen nutrition index (NNI) has been introduced (Justes et al., 1994; Lemaire et al., 1989), which is the ratio between the actual N% to the optimal (critical) N% value. Hereby, knowledge of plant dry biomass is required. If NNI > 1.0, the crop is provided with sufficient (or excess) N. In contrast, NNI < 1.0 implies N deficiency, leading to crop stress. Considering this dilution phenomenon of N%, Baret et al. (2007) proposed to use Narea to calculate NNI instead of N%, since the resulting N excess or N deficiency can be directly quantified (see also Fig. 4 in Baret et al. (2007)).

In summary, it appears that none of these review studies directly examined the potential of protein content as proxy for nitrogen but rather proposed the use of the relation between N and chlorophyll content for nitrogen retrieval. A reason for this may be the lack of satellite imaging spectroscopy missions in the past that could provide time series of narrowband reflectance data covering also the SWIR spectral region, which allows the monitoring of spectral features of proteins. Another explanation may be that the impact of proteins on canopy reflectance is rather small compared to other parameters by contributing with only 2–4% to the total variation of canopy reflectance (Wang et al., 2018). It may therefore be more straightforward to estimate chlorophyll content exhibiting a distinctive signal in the visible spectral domain. Most of the studies highlight the meaningfulness of empirical methods for N retrieval (Clevers and Kooistra, 2012; Homolová et al., 2013), whereas deterministic approaches such as radiative transfer modelling of proteins is still in its infancy. Nevertheless, the physically-based N retrieval would be the only option for a transferable monitoring system within the agricultural context. With our review study we want to stress on these discussions and gaps, and point toward possible new developments.

4 Algorithms for nitrogen retrieval from remote sensing data

Thomas and Oerther (1972) made pioneering efforts to estimate N from remote sensing and were the first exploiting the relationship between foliar N concentration and laboratorybased leaf reflectance measurements of sweet pepper plants. Since then, methodologies for remotely sensed retrieval of N in the agricultural context were further developed and improved. In the years 1991–1992, the NASA Accelerated Canopy Chemistry Program (ACCP, 1994) was established to identify a sound theoretical and empirical basis to estimate N (and lignin) concentrations in vegetation from remotely sensed imagery.

From the controlled conditions of laboratory-based reflectance measurements to field-, airborne and satellite-based studies, a number of effects potentially increase uncertainties of N retrieval. According to Malenovský et al. (2019) three main categories of variability and uncertainty sources can be identified in the context of biochemical and biophysical plant traits retrieval from (satellite) imaging spectroscopy data. The first category includes the errors referred to the sensor, including sensor calibrations, and radiometric, geometrical and atmospheric corrections (Baret and Buis, 2008; Gao et al., 2009). The second category of uncertainties arises from the parameterizations and assumptions specific to retrieval algorithms (Knyazikhin et al., 2013b). The third group involves the acquisition of in situ field data which are the basis for physically quantifying the uncertainty of the derived vegetation traits (Richter et al., 2012a). Our review mainly focuses on the second category and investigates uncertainties, problems and challenges associated with the different methods for the estimation of N.

4.1 Parametric regressions methods

In the last decades, predominantly simple empirical algorithms based on parametric regressions have been used for N estimation from hyperspectral signatures in the agricultural context. These approaches are by far the most applied and largest group of biochemical and biophysical vegetation trait estimation techniques. Parametric regression models include narrowband vegetation indices (VI) exploiting wavelengths mainly in the VIS, red edge, NIR, but also SWIR, e.g. (Chen et al., 2010; Clevers and Kooistra, 2012; Eitel et al., 2008; He et al., 2016; Herrmann et al., 2010; Jay et al., 2017; Moharana and Dutta, 2016; Stroppiana et al., 2009a). Several studies have demonstrated improved retrievals of canopy N when protein-related absorption bands in the NIR and SWIR were incorporated into narrowband indices (Ferwerda et al., 2005; Herrmann et al., 2010; Serrano et al., 2002).

4.2 Linear nonparametric regression methods

Linear nonparametric regression methods, also known as "chemometrics", were also exploited for the estimation of N. Chemometrics can be summarized as data-driven approaches using mathematical and statistical techniques to extract chemical and physical information from complex data, such as those measured from hyperspectral sensors. In contrast to the statistical linear parametric approaches, chemometrics do not reduce the spectral information to defined indices but rather add value to the data by transforming them into new features (Davies, 2012): principal component analysis (PCA) is a full-spectrum chemometric procedure that converts a set of observations of possibly correlated variables into a set of principal components of linearly uncorrelated variables by applying orthogonal transformation (Liu et al., 2017). In hyperspectral remote sensing, PCA methods are usually applied for spectral dimensionality reduction (Rivera-Caicedo et al., 2017). The principal components can then be used to calculate regression coefficients for the derivation of biophysical or biochemical vegetation traits (=Principal Component Regression, PCR). Stepwise multiple linear regression (SMLR) uses a defined set of wavebands as independent (predictive) variables (Grossman et al., 1996). Within each step another predictive variable is added or subtracted from the set of explanatory variables in the function based on specific criteria, such as the Akaike information criterion (Yamashita et al., 2007). The full-spectrum methods of "partial least squares regression" (PLSR) have been widely used

in chemometrics (Atzberger et al., 2010; Wold et al., 2001). This technique relates two data matrices (i.e. reflectance and corresponding biochemical trait) by a linear multivariate model taking also the structure of these matrices into account. With this characteristic, PLSR goes beyond traditional regression (Wold et al., 2001), transforming the spectral feature space in a way that the resulting factors account for a maximum of variance in the covariance with the target variable(s). PLSR is able to analyze a vast amount of noisy, collinear data, and its precision increases with the increasing number of relevant variables and observations.

4.3 Nonlinear nonparametric regression methods

Further, nonlinear nonparametric regression methods, namely machine learning regression algorithms, have been adopted for vegetation remote sensing (Verrelst et al., 2019). The group of machine learning includes kernel methods such as support vector regression (SVR), e.g. described in Smola and Schölkopf (2004), a kernel-based regularized regression algorithm. One of today's most promising machine learning methods is Gaussian processes regression (GPR), developed by Rasmussen and Williams (2006). These types of machine learning models have excelled in Earth observation problems in recent years and were mainly introduced for model inversion and emulation of complex codes (Camps-Valls et al., 2016). GPR belongs to kernel based methods relying on a solid Bayesian formalism and thus allow for a formal treatment of uncertainty quantification and error propagation (Camps-Valls et al., 2019; Verrelst et al., 2012). As a probabilistic approach, a GPR model is trained to find individual functions to all training data by fitting a mean and a covariance function. Other examples of machine learning include random forest regression (RFR) and artificial neural networks (ANN). These methods have become increasingly popular in image processing for numerous applications, which is due to their reliable performance and robustness (Verrelst et al., 2012). Random forest regression is a data mining method developed by Breiman (2001) combining a large set of decision trees. It is a robust ensemble learning technique able to handle large numbers of input variables. The widely used approach of ANNs makes use of error backpropagation through the network of perceptrons (Waske et al., 2009). A disadvantage of ANNs is that they are prone to overfitting, meaning that they strongly depend on input quality and may struggle with unknown values outside the range in which they were trained (Kimes et al., 2000). Problems of underfitting and in particular - overfitting are, however, generally applicable to machine learning models. Overfitting occurs when the models learn random fluctuations and noise from the training data in high detail. Performance and generalization capability of the trained models is adversely affected if these features are not present in the validation data sets. Underfitting implies that the model did not capture essential trends in the training process and thus cannot be applied to new data. Alternative machine learning methods could be tested or hyperparameters tuned to improve performance (Verrelst et al., 2019).

4.4 Physically-based methods

As an interesting alternative, physically-based methods using radiative transfer modelling (RTM) offer the advantage of robustness and transferability (Berger et al., 2018a; Kimes et al., 2000), but have not been fully examined for the assessment of crop N (Wang et al., 2015a). Physical modelling of the spectral signal of leaf N content has been discussed controversially. The authors of the original leaf optical properties model PROSPECT

(Jacquemoud and Baret, 1990) - a radiative transfer model based on Allen's generalized "plate model" simulating the optical properties of plant leaves from 400 to 2500 nm based on a limited set of biochemical constituents and a structure parameter – were the first who tested the implementation of protein (as surrogate of N) into the absorption and scattering processes of the model. These and later attempts were abandoned due to the strong covariance of nitrogen with other N-containing compounds rendering the results inconsistent in particular for fresh leaves (Baret and Fourty, 1997; Fourty et al., 1996; Jacquemoud et al., 1996; Kokaly et al., 2009). Wang et al. (2015b) incorporated specific protein as well as cellulose + lignin absorption coefficients into PROSPECT-5 (Feret et al., 2008). In their work, the authors demonstrated the applicability of the leaf optical properties model of separating specific absorption coefficients (SAC) for proteins and successfully evaluated the feasibility of estimating leaf protein content by means of model inversion from fresh leaf spectra in the spectral range of 800–2500 nm. This was possible through recalibration of the SACs using a new calibration algorithm (Feret et al., 2008; Wang et al., 2015b). The approach was applied by Berger et al. (2018b) to an agricultural area using a look-up table (LUT) based inversion of the actual PROSPECT model PROSPECT-D (Féret et al., 2017) adapting the SACs of protein content for fresh leaves from Wang et al. (2018), and coupling it with the actual version of Scattering by Arbitrary Inclined Leaves (SAIL) model "4SAIL" (Verhoef et al., 2007). The SAIL model (Verhoef, 1984) is one of the earliest and to date probably the most popular canopy bidirectional reflectance model (Jacquemoud et al., 2009). SAIL is based on the one dimensional (1-D) model developed by Suits (1972) to simulate the bidirectional reflectance factor of plant canopies which are assumed as turbid medium. Scattering and absorption is simulated by four upward and downward radiative fluxes. The coupling of SAIL and PROSPECT corresponds to the PROSAIL model (Jacquemoud et al., 2009). As recently reviewed by Berger et al. (2018a), PROSAIL has been used for almost 30 years within hundreds of studies in forward and inverse modes for numerous vegetation-related applications. The afore mentioned LUTbased inversion technique consists of computing a cost function - most authors choose the root mean squared error (RMSE) cost function (Danner et al., 2017) - and finding its minimum over a (PROSAIL) data base of simulated spectral signals. The solution is often the mean or median of all spectra within a defined interval, such as 10–20% of the lowest RMSE values. The final solution for the estimated traits corresponds to the mean (or median) of the respective model input parameters (LAI, chlorophyll content, etc.). LUT-based approaches have been applied by numerous studies over the solar domain (Atzberger and Richter, 2012; Danner et al., 2017; Richter and Timmermans, 2009; Wang et al., 2015b; Weiss et al., 2000). Adapted PROSPECT models, e.g. by Wang et al. (2018) and Berger et al. (2018b), provide promising tools to estimate leaf proteins and thus could be further exploited for N retrieval application studies in an agricultural context (actual versions down-loadable from https://artmotoolbox.com/). Scaling up to the canopy level can be performed by implementing a protein-simulating PROSPECT version for instance into the Soil-Canopy-Observation of Photosynthesis and Energy fluxes (SCOPE) model (van der Tol et al., 2009) or coupling it with the Discrete Anisotropic Radiative Transfer (DART) model (Malenovský et al., 2013). The fact that protein absorption features are partly overlaid by absorption of other vegetation properties, such as water content, leads to difficulties by using simple parametric regression approaches for the retrieval (Feret et al., 2018). Thus,

RTMs should be specifically exploited to decouple confounding factors in the estimations of N or proteins, such as water content or canopy structure. The location of the specific protein absorption features (see Table 1) emphasizes the importance of incorporating the SWIR spectral region for RTM-based retrieval applications. Still, the ill-posed inverse problem may occur for which many solutions have been proposed, such as specific merit functions (Wang et al., 2015b) or the object-based retrieval (Atzberger and Richter, 2012). An exhaustive review to this and related topics is given in Verrelst et al. (2015).

4.5 Hybrid methods

Using RTMs, for instance PROSAIL, in combination with machine learning methods leads to hybrid retrieval procedures. Hybrid models offer the advantage of a physical basis provided by RTMs and the computational efficiency and flexible procedure provided by regression methods. While physical models explore the underlying causality between inputs and outputs, machine learning algorithms are able to isolate relevant inputs for a given output from large datasets (Baker et al., 2018). Practically, the machine learning model is trained on a simulated RTM database, thus rendering the need for in situ reference data obsolete during the training stage (Verrelst et al., 2019). The estimation of several biophysical and biochemical variables has been successfully performed with hybrid inversions: for instance leaf area index (LAI) retrieval by combining PROSAIL with ANNs (Verger et al., 2011), and chlorophyll content retrieval combining PROSAIL with RFR models (Doktor et al., 2014).

Table 2 lists the discussed main retrieval methods for N sensing from hyperspectral data within an agricultural context.

5 Satellite imaging spectroscopy missions for N sensing

In the past few decades, remote sensing has become an attractive technique to estimate crop N at the field scale, i.e. agricultural farmland within at least decametric spatial resolutions. In this context, field spectroscopy has been mainly used for biochemical characterization of vegetation during ground based campaigns, for instance by Alchanatis et al. (2005); Clevers and Kooistra (2012); Ecarnot et al. (2013); Li et al. (2010); Thorp et al. (2012). Proximal sensing at field level is locally constrained and often time-consuming. Instead, the monitoring of large areas using imaging spectroscopy has demonstrated advantages for ecological studies in the domains of agriculture, food security, soils, biodiversity, environmental degradation and hazards, hydrology, forestry and others, as reviewed and discussed by Rast and Painter (2019) or Ustin et al. (2004). For agricultural areas, satellites can provide spatially and temporally consistent information over the growing season capturing the biological lifecycle of the crop and spatial heterogeneity of every field on a weekly basis and at low cost (Atzberger, 2013). This is not feasible with field-based spectroscopy, unmanned aerial vehicles (UAV) or airborne sensors, which usually only fly within organized campaigns. For years, there were only two imaging spectrometers in orbit - Hyperion on NASA's Earth Observing-1 satellite and Compact High Resolution Imaging Spectrometer (CHRIS) on ESA's Proba-1. Yet, both sensors were not designed for global mapping. Hyperion data were for instance used by Abdel-Rahman et al. (2013) to estimate

sugarcane N concentration by means of a RFR. CHRIS data were exploited by Castaldi et al. (2016) to estimate the grain N uptake of wheat by means of stepwise regression combined with stepwise variance inflation factors (VIFs) analysis and linear mixed effect model (LMEM). Marshall and Thenkabail (2015) compared the estimation of crop biomass as important indicator for N uptake using spectral indices derived from multi-spectral sensor data (IKONOS, GeoEye-1, Landsat ETM+, MODIS, WorldView-2) with narrowband indices from the EO-1 Hyperion sensor. Results demonstrated that hyperspectral narrowband indices could explain 5–31% greater variability of biomass than broadband indices. This finding strengthens the need for satellite data with higher spectral resolution.

The rare usage of satellite imaging spectroscopy will change in the upcoming years with the evolution of several satellite missions that will provide frequent observations of large areas. Thus, a key role for scientists within the field of spectroscopy in optical remote sensing is the challenge of upscaling the developed models from field to spaceborne sensors (Mutanga et al., 2015). An actual overview of hyperspectral spaceborne systems under development, in preparation and recently launched is given by Rast and Painter (2019). Table 3 presents an overview of recently launched and planned spaceborne imaging spectrometer missions, which could be of interest for protein and N retrieval. These missions of interest include for instance the Italian PRISMA (PRecursore IperSpettrale della Missione Applicativa) satellite (Loizzo et al., 2019), launched on 22nd March 2019. Another example is the Hyperspectral Imager Suite (HISUI), developed by Japanese Ministry of Economy, Trade, and Industry (METI), which was launched on 5th December 2019 (Matsunaga et al., 2017). Shortly after, the German Environmental Mapping and Analysis Program (EnMAP) is foreseen for launch in April 2021 (Guanter et al., 2015). It will acquire information over 30-km-wide areas in the across-track direction, with 30 m ground sampling distance, and measure the spectral domain from 400 nm to 2500 nm over 242 spectral bands with a SSD from 6.5 to 10 nm. Examples of other missions are, among others, the Italian-Israeli Spaceborne Hyperspectral Applicative Land and Ocean Mission SHALOM, envisaged for 2022 (Natale et al., 2013). Some missions are still under investigation, as for instance the former Hyperspectral InfraRed Imager (HyspIRI) mission (Lee et al., 2015), now renamed to NASA's Surface Biology and Geology (SBG), and the high-priority mission candidate Copernicus Hyperspectral Imaging Mission for the Environment (CHIME), described in Nieke and Rast (2018).

The instrumental characteristics of PRISMA, EnMAP, CHIME and similar systems (see Table 3) are very auspicious for N retrieval studies: they cover the SWIR spectral domain where main protein (nitrogen) absorption features are located. The study of Pellissier et al. (2015), for instance, successfully estimated foliar N concentration using a PLSR approach, and concluded that the proposed HyspIRI (SBG) mission is promising for detecting canopy N concentration across multiple forms of managed grasslands. The contiguous sampling of spectral bands by these sensors allows the discrimination of leaf biochemical components such as proteins. It would be optimal if the spatial resolution provided by the future sensors is in accordance with ESA Sentinel-2 sensors (10–20 m) to allow the mapping of small agricultural fields. Nonetheless, a ground sampling distance of 30 m is still acceptable for many agricultural regions and applications (Rast and Painter, 2019). Thus, with all those missions ahead, an era of operational agricultural monitoring systems is about to

start providing for the first time hyper-spectral time series over large areas. This will enable the further exploitation and validation of efficient retrieval algorithms for vegetation biochemical plant constituents such as nitrogen (Hank et al., 2019). Moreover, time series of crop N content are crucial for the management of fertilization timing to be performed at specific growth stages (Weiss et al., 2020). Nevertheless, the availability of these new hyperspectral data streams will also pose a variety of challenges, such as the correction of atmospheric effects (Gao et al., 2009), temporal gap filling (Gao et al., 2008), understanding of floristically mixed pixels (Ollinger, 2010) or uncertainties arising from leaf-to-canopy upscaling approaches (Malenovský et al., 2019). Given the relatively large pixel size of these new and upcoming hyperspectral satellite sensors (with GSD of 10 m to 30 m, see Table 3), the measured signals will always represent a heterogeneous surface composite; meaning that one pixels' signal is the result of the interactions of electromagnetic radiation with several (plant and soil) constituents (Keshava and Mustard, 2002). Therefore, specific techniques, such as Spectral Mixture Analysis (SMA) were explored, as it was done for instance by Lelong et al. (1998) for stress detection and mapping of wheat agronomic variables. A review of available SMA methods and results of endmember variability reduction in SMA is provided by Somers et al. (2011). The handling of large data volumes will also be an issue, but should become feasible in the near future due to the broad availability of artificial intelligence, machine learning and advanced data analytics (Rast and Painter, 2019).

6 Literature review of nitrogen studies for agricultural applications

6.1 Identification and screening of relevant literature

A systematic literature review was carried out including identification, screening, eligibility and inclusion of relevant records, described in the scheme of the Supplementary material (Supplement 1). The ISI Web of knowledge (Gillis, 2018) and ScienceDirect search engines were the main sources for the identification of relevant records. The topics "hyperspectral" or "imaging spectroscopy", and "nitrogen concentration" or "nitrogen content", and "agriculture" were searched in all variations for the period of 1994–2019. After collecting all apparently relevant records (number of studies, n = 426), a first screening process was performed to remove duplicates and references not relevant for the review study, such as conference proceedings or other non-peer reviewed works or reports, resulting in n = 391 papers for further investigation. In the next step, titles and abstracts screening was performed to exclude all studies from other research fields. In the eligibility part, the remaining n = 167 records underwent intensive analysis. We specifically excluded studies with the following criteria:

- No direct estimation of N (but chlorophyll content, nicotine, yield, proteins, fiber quality or other proxies);
- (2) N of soil was determined;
- (3) Forest, shrubs, wetlands or other vegetation types instead of agriculture;
- (4) Resampling into multi (super-) spectral data (e.g. Quickbird, Sentinel-2);
- (5) Only rough classification of N supply without concrete measurements;

(6) Review study.

Finally, a total number of 125 records (see Supplement 2) were identified fulfilling the main criterion of crop N estimation with different methods from hyperspectral or imaging spectroscopy data within an agricultural context. From those studies relevant information according to the following criteria were extracted:

- Spectral exploitation: sensor type, sensor with respective spectral domain;
- Retrieval method;
- Usage of terms (N content/concentration or others);
- Geographic location of the experiments;
- Crop types and scale (leaf and/or canopy).

Fig. 3 shows the absolute number of N retrieval studies extracted from our meta-review from 1994 to 2018 that used data from hyper-spectral sensors to estimate N within an agricultural context. This trend also reflects the general trend of remote sensing application studies, as shown for instance by other reviews in the context of agriculture (Berger et al., 2018a; Weiss et al., 2020).

6.2 Sensor systems

Concerning the usage of different sensor systems, it can be observed that predominantly proximal sensing using field spectrometers has been carried out to acquire canopy spectral reflectance in the field (Fig. 4), as it was the case in 75% of all studies. The most used instruments were from the Analytical Spectral Devices (ASD) series, operating in the spectral range from 350 to 2500 nm, used for instance by Stroppiana et al. (2009b); Wang et al. (2013); Yao et al. (2014). Field-based measurements are followed by airborne acquisitions, for instance with Compact Airborne Spectrographs Imager, CASI (Boegh et al., 2002; Chen et al., 2010), AISA Eagle (Nigon et al., 2015) and AVIRIS (Perry and Roberts, 2008). The low use of satellite-based data, mainly from CHRIS/Proba and Hyperion, was already discussed in Section 5. A few studies used laboratory spectroscopy for leaf N simulation (Chen et al., 2019) or additionally simulated reflectance from hyperspectral sensors by means of radiative transfer modelling (Eitel et al., 2008). UAV or drones, offering novel remote sensing tools for precision agriculture, were used for instance by Näsi et al. (2018). All studies cited in this section belong to the list of identified studies (see Supplement 2).

6.3 Retrieval methods

Retrieval methods for crop N used by the identified studies can be grouped into six general approaches, similar to the state-of-the-art retrieval methods categorization of Verrelst et al. (2019), see also Table 2:

- Parametric regression methods, involving the relation of N to narrow spectral bands or derivates, ratio narrowband vegetation indices (NB-VI) and spectral shapes or transformations (Section 6.3.1);
- Linear nonparametric regression methods or chemometrics (Section 6.3.2);

- Nonlinear nonparametric regression methods or machine learning regression algorithms (Section 6.3.3);
- Physically-based or radiative transfer models (RTM) (Section 6.3.4);
- Alternative data sources such as sun-induced fluorescence based methods (SIF) (Section 6.3.5);
- Hybrid or combined techniques (Section 6.3.6).

Fig. 5 gives an overview of the frequency of applied methods within their categories of the n = 125 studies. Note that many of the studies used several approaches; hence the sum of all applied algorithms does not correspond to the total number of studies.

6.3.1 Application of parametric regression methods—The parametric regression models were divided into three main subgroups. Note that further subdivision is possible but exceeds the scope of this study. Fig. 5 indicates nine studies using the correlation of spectral bands or derivatives to estimate N. Tian et al. (2011), for instance, exploited the relationships between the red edge positions (REP) derived from different algorithms and canopy leaf N concentration of field-grown rice. The REP, defined as the position of the sharp change in leaf reflectance between 680 and 750 nm, is mainly related to changes in chlorophyll content (Horler et al., 1983) and thus secondarily to N.

The overall most used approaches with 62% of all identified records were the narrowband ratio vegetation indices. Only few studies pointed out the limitations of the indices-based approaches, such as limited transferability to different sites. For instance, Moharana and Dutta (2016) established regression models for N indices from Hyperion imagery in an agricultural system. The authors found that Simple Ratio (SR) and leaf N concentration indices, exhibiting linear and nonlinear relationships, were completely different from those developed by other studies. Camino et al. (2018) pointed to the importance of SWIR spectral information. They could show that nitrogen indices (NIs) centered at 1510 nm obtained better agreement with N% ($R^2 = 0.69$) than traditional chlorophyll indices (TCARI/OSAVI $R^2 = 0.45$) and structural indices (NDVI $R^2 = 0.57$), which are solely based on the VNIR region. Nevertheless, the major part of the studies in this category estimated plant N on the basis of its correlation to chlorophyll content.

In summary, although these approaches delivered acceptable to good accuracies of N estimation and are fast in calculation and prediction times, it remains questionable whether the full complexity of hyperspectral data observed by a spectroradiometer can be captured by means of transformed data originating from a few discrete spectral bands (Verrelst et al., 2019). The main drawbacks of VI-based retrievals can be summarized in five points according to Atzberger et al. (2011):

(1) The ill-posedness of the inverse problem occurs, meaning that the resulting value of combining two bands can be a result of several different parameter solutions.

- (2) By using only few spectral bands, available hyperspectral information is only partially exploited, and enhanced sensitivity to noise may occur if narrow bands of low signal-to-noise (SNR) were combined.
- (3) Sensor-specific calibration may not be transferred to other sensors due to different instrumental characteristics.
- (4) Overspecialization or over-fitting can occur, meaning that the established VI-N models are specific to the dataset used for model calibration. This includes sensitivity to crop type, growth stage, underlying soil type, sun-sensorconstellation, time and date.
- (5) Finally, the need to collect in situ reference data for model establishment requires man power, availability of instruments for all acquisitions and campaigns organizational efforts.

Some of these drawbacks can be addressed by generating synthetic databases with appropriate RTMs (Hunt et al., 2013), or by the exploitation of hyperspectral reflectance data collected during intensive campaigns in diversified agricultural regions (Thenkabail et al., 2013b).

6.3.2 Use of chemometrics—PLSR proved to be the most popular technique in the group of chemometrics for studying N from hyperspectral data, and follows directly the NB-VI regarding all methods. It was employed by 37% of all studies selected in the review process. This can be explained, among other reasons, by the fact that the classical PLSR method is available in many software packages and is straightforward in implementation. Worth to mention in this context is the highly cited study of Hansen and Schjoerring (2003) who demonstrated that PLSR improved the prediction of N concentration and decreased the root mean square error (RMSE) by 24%, compared to the application of narrowband indices. The authors specifically highlighted that PLSR analysis "...may provide a useful exploratory and predictive tool when applied on hyperspectral reflectance data". The frequent use of PLSR by N studies mirrors its advantage of combining the positive features of PCA and SMLR, as hereby the covariance to the biochemical traits of interest is considered. One should carefully choose the number of components; otherwise overfitting caused by too many components may occur (Hansen and Schjoerring, 2003). Inoue et al. (2012) stated that PLSR may be more suitable than principal component regression (PCR) for predictive purposes since the resulting latent variables (LV) of a PCR account only for variance in the feature space instead of the covariance between the independent and target variables as in case of PLSR. Nevertheless, also the PLSR method may not always provide optimal results due to possible disturbance by certain wavebands (Inoue et al., 2012). Some optimized PLSR methods have been developed to overcome these limitations, such as interval PLSR (Nørgaard et al., 2000).

Stepwise multiple linear regression was applied by Miphokasap et al. (2012) who found higher correlations with N content than by models based on narrowband vegetation indices. The authors illustrated their findings with the utilization of the full spectral domain of the SMLR (400 to 2500 nm). The benefit of SWIR bands was hereby demonstrated by the study of Dunn et al. (2016) who identified the four most important wavelengths for

predicting N uptake by means of a new calibration created by multiple linear regression using 738 nm, 1362 nm, 1835 nm and 1859 nm spectral bands. Since (S)MLR approaches do not compress spectral data to extract most essential information, sensor noise has little chance to be canceled out (Jacquemoud et al., 1995). An advantage of SMLR is that one can focus on spectral regions of interest when the relation to the variable is well-known (Fourty and Baret, 1997). Moreover, SMLR methods may deliver high predictive accuracies but sometimes at the cost of overfitting and large computational loads, in particular when excessive numbers of variables are selected (Yu et al., 2013; Zhou et al., 2018).

All in all, chemometric methods may be able to provide relatively high retrieval accuracies of N estimation. These methods may be also more suitable for N sensing from hyperspectral data as the NB-VI or other linear parametric approaches due to the above discussed advantages of exploiting the full available spectrum. However, with this approach the local calibration and validation limits the transfer of the models to another space and time (Verrelst et al., 2019). A major advantage of the two full spectrum methods PCR and PLSR is that they are able to overcome multicollinearity, which is the inherent problem of (S) MLR (Grossman et al., 1996). Multiple linear regression methods cannot cope with multicollinearity of hyperspectral data and therefore they tend to be unstable which limits their predictive capabilities (Inoue et al., 2018). It also cannot be assured that established components contain the most relevant information of the biochemical property of interest. Regarding computational efficiency, chemometric techniques may be in slight advantage over more complex machine learning regression and in particular compared to RTM-based approaches. Although machine learning models may be more complex to implement than PLSR approaches, they can deliver explicit estimates of uncertainties and can provide higher mapping qualities as found by several studies (Ashourloo et al., 2016; Verrelst et al., 2019). These are important aspects to be considered in view of future operational processing of vegetation properties from spaceborne imaging spectrometer data streams.

6.3.3 Use of machine learning regression algorithms—Regarding the usage of nonlinear nonparametric regression methods, random forest regression, artificial neural network and support vector regression have been applied in almost equal frequency (RFR, n = 10; ANN, n = 8; SVR, n = 9). Pullanagari et al. (2016) compared different nonparametric models (including PLSR, SVR and RFR) regarding their performance to estimate N% of heterogeneous mixed pasture and found that RFR yielded the highest accuracy. Liang et al. (2018) also reported that RFR algorithms largely improved the prediction of N content compared to two VI-models. Artificial neural networks were applied for instance by Yi et al. (2010), who used the advantages of a PCA with an ANN approach and successfully determined N concentration of rice from spectral reflectance. Miphokasap and Wannasiri (2018) estimated N% of sugarcanes: their results indicated higher correlation coefficients using SVR-based Radial Basis Function (RBF) kernel compared to SMLR models. A large comparative study of Yao et al. (2015) evaluated the performance of six different algorithms, including continuum removal, VI, SMLR, PLSR, ANN and SVR to estimate N concentration in winter wheat. Results indicated that all nonparametric methods performed similar with SVR being the most flexible approach regarding the handling of potential confounding factors.

GPR models, in contrast, have only been evaluated in one study of Zhou et al. (2018) who specifically analyzed the influence of spatial resolution on the separation of leaf N signals from the background in rice canopies. Comparing the results of the GPR model with PLSR and VIs, the authors demonstrated that GPR achieved the best prediction of leaf N concentration.

These comparative studies demonstrated that machine learning often yielded higher estimation accuracies than linear nonparametric approaches, mainly for N% predictions (Miphokasap and Wannasiri, 2018; Pullanagari et al., 2016; Yao et al., 2015; Zhou et al., 2018).

Although these studies have demonstrated the feasibility and applicability of machine learning methods for estimating N of different crops, the models have been exclusively trained and calibrated on in situ measured field data as it is the case for chemometrics and parametric regressions. Therefore, it is difficult to physically inter-compare the results using varying spatial resolutions, diverse sensors and model configurations. Regarding prediction time, nonlinear nonparametric regression algorithms can map biochemical and biophysical vegetation traits quasi instantaneously from hyperspectral scenes, in particular when combined with dimensionality reduction such as PCA. Among these machine learning methods, random forests are fastest in training and prediction; they are also less prone to overfitting and can be regarded as robust to outliers and noise (Belgiu and Dr gu, 2016). The flexibility offered by kernel methods, in particular the GPRs, make them the most interesting candidates to solve nonlinear problems in hyper-spectral remote sensing. These outstanding methods convince with their intrinsic regularization capabilities and low sensitivity to data dimensionality over other methods for vegetation traits retrieval (Tuia et al., 2018). Due to their capability to provide uncertainty intervals of the predictions, GPR methods will certainly play a major role for automatic processing of biochemical vegetation traits mapping in the future, in particular within hybrid retrieval schemes.

6.3.4 Use of radiative transfer models—Only three studies applied a radiative transfer model for N retrieval in an agricultural context: the N-PROSPECT model (Z. Li et al., 2018; Li et al., 2019; Yang et al., 2015). N-PROSPECT is a modified PROSPECT model version, where the SACs corresponding to chlorophylls were replaced with SACs corresponding to N in order to simulate leaf nitrogen density (LND, in μ g/cm²) at the leaf scale (Yang et al., 2015). In the study of D. Li et al. (2018), crop N status both at leaf and canopy scales could be retrieved through coupling of N-PROSPECT with the SAIL model (N-PROSAIL).

The N-PROSPECT model, however, was restricted to winter wheat since it was calibrated on four winter wheat cultivars. Hence it is only suitable for estimating N content with experimental set ups similar to the calibration situation (Yang et al., 2015). Finally, this model relies on the correlation between leaf chlorophyll content and N, for which limitations have been highlighted in the previous sections.

Researchers from the radiative transfer community are less optimistic that proteins (nitrogen) can be retrieved by means of physically-based methods (Homolová et al., 2013;

Jacquemoud et al., 1996; Knyazikhin et al., 2013a). This is also mirrored by our literature review with only three studies having used RTM-based approaches within the agricultural context to derive N. It is therefore questionable if satellite-based reflectance will allow for an accurate, transferable and physically-based retrieval of N. According to a broad body of research, the accurate estimation of N is mainly possible through indirect correlations with those traits that are more explicitly linked to the radiative transfer - mostly chlorophyll content. Protein content should be the trait of choice according to its strong connection to N. To guarantee transferability and robustness of monitoring systems, these mechanistic approaches are urgently needed. Large training data sets can be established for machine learning models using deductive capabilities of these physical models to extrapolate to predictions not present in collected in situ data in the past (Baker et al., 2018). This shall motivate researchers to further exploit RTMs tailored to simulate the absorption features of proteins for the retrieval of N.

6.3.5 Alternative data sources: SIF—A few studies used sun-induced chlorophyll fluorescence (SIF) as alternative data source to estimate crop N. Jia et al. (2018) assessed the feasibility of quantifying leaf N concentration from SIF based on the response of SIF to chlorophyll content. The authors found that downward SIF yield indices gave more accurate estimation than the upward SIF yield (FY) indices, though measurement with today's available instruments is still difficult. In the study of Camino et al. (2018) a higher accuracy of N retrieval could be achieved when chlorophyll fluorescence was included compared to models only built based on chlorophyll a + b (C_{ab}), dry matter (C_m) or equivalent water thickness (Cw). The contribution of SIF to N estimation may be highly valuable, though only the chlorophyll-based relation to N is taken into account. Ground-based measurement and modelling studies have reported about the suitability of red and far-red SIF for the detection of N deficiency, which affected many parts of the photosynthetic apparatus, including photosynthetic pigments, thylakoid proteins, and the soluble enzymes such as rubisco (A et al., 2015; Mohammed et al., 2019). In this review, however, our focus is on the N retrieval studies using reflectance measurements from hyperspectral sensors but not on SIF measurements.

6.3.6 Implementation of hybrid or combined methods—Hybrid techniques denominate a combination of at least two methods in synergic use. In biophysical and biochemical variable retrieval studies, this method is often the combination of machine learning methods and RTMs, thus combining the generalization level of a physically-based method with a flexible and computationally efficient nonlinear nonparametric statistical method (Verrelst et al., 2019). This kind of studies were not found within the identified records which may be explained by the lack of publicity available RTMs involving N (protein) modelling. We thus extend the definition to include studies using different kinds of methods in combination, for instance crop growth and canopy reflectance modelling. As such, Thorp et al. (2012) combined the DSSAT Cropping System Model (CSM) with the PROSAIL radiative transfer model (CSM-PROSAIL). This study also used the relationship between chlorophyll and N content, by applying a conversion factor to calculate the CSM nitrogen state variable (g cm⁻²) from C_{ab} (μ g cm⁻²).

The additional information of SIF can be exploited in synergy with RTM-based variable retrieval: in Camino et al. (2018) multiple regression models were established to estimate N with and without including airborne retrieved sun-induced fluorescence. The model performance could significantly be increased by combining SIF with PROSAIL derived variables in the regression.

Fig. 6 shows the development of the application of different method-categories over time. In addition to an overall increasing trend, an increasing interest in nonlinear nonparametric regression algorithms and hybrid/combined methods is obvious. There is also a positive trend in the usage of RTMs in view of the slow but steady increase of physically-based modelling of N content (Li et al., 2019; Wang et al., 2015a; Wang et al., 2015b; Yang et al., 2015).

6.4 Use of technical terminology

There is a lack of consistency regarding the terminology within the identified records. Studies estimating nitrogen from remote sensing acquisitions use diverse expressions, for instance N content, N concentration, N uptake, N status, N rate, N accumulation, N density or even only refer to "N" without any unit or additional designation. To enable a valid comparison between nitrogen-related studies, a standardization of terms would be highly desirable. We recommend that the definition and units of nitrogen should be clearly stated in future works.

Moreover, regarding retrieved nitrogen information from remote sensing, we strongly recommend to distinguish between two measures (D. Li et al., 2018; Scott Green et al., 2003; Wang et al., 2015a):

- Area-based measure "nitrogen content" (N_{area}), expressed per surface unit, and e.g. given in [g·cm⁻²] at leaf scale, in [g·m²] at canopy scale, or [kg·ha⁻¹] at field scale.
- Mass-based measure "nitrogen concentration" (N%), expressed per leaf mass unit (or other plant organs), usually using dry weight, and e.g. given in [mg·g⁻¹] or [%].

The leaf dry mass per unit leaf area (LMA) can be used to convert N % to N_{area} (Baret and Fourty, 1997):

$$N_{\text{area}} = N\% \cdot LMA \tag{1}$$

Hence, plant N content, expressed on a land area basis, is the product of plant N concentration and plant dry biomass in kg ha^{-1} (Chen et al., 2010).

 N_{area} is important from a physiological perspective since it is highly correlated to the photosynthetic capacity of leaves and thus to carbon fixation (Evans, 1989; Rosati et al., 2000).

Narea is also considered as the candidate trait with the strongest relationship to V_{cmax} and J_{max} (Dechant et al., 2017; Kattge et al., 2009). Expressing the light saturated photosynthetic rate and nitrogen on the basis of leaf dry weight (i.e., N%), this species-independent relationship clusters around a straight line and reaches zero at 0.6 mmol N g⁻¹ (Evans, 1989). Osnas et al. (2013) found that leaf traits were primarily proportional to leaf area, but not to leaf mass as suggested by Wright et al. (2004). Moreover, reflectance and transmittance are not directly sensitive to the concentration, but rather to the content of absorbers per unit leaf area. Therefore, Narea should be estimated and - if needed converted into N% using LMA (Baret and Fourty, 1997). The inverse linear relationship of biomass and N% (dilution phenomenon) shows that with increasing growth stage from planting to maturity, crop N concentration decreases while biomass and LAI increase (Chen et al., 2010; Plenet and Lemaire, 1999). Thus, the usage of N content is more useful and suitable to describe optimal N status (Baret et al., 2007). Moreover, as proposed by Kattenborn et al. (2019) for pigments, the area-based measures are more suitable for upscaling purposes from leaf to canopy level. Upscaling leaf Narea to aboveground Narea only requires an estimate about total foliage (e.g. LAI) and not, as with N%, additional knowledge about dry matter content.

Nevertheless, the identified records of our meta-analysis focused mainly on N%, see also Fig. 7. In contrast, the estimation of the area-based measure, N_{area} , has been rarely performed, as it was also pointed out by D. Li et al. (2018). Unfortunately, confusion about terminology is present, since many studies use the terms content and concentration interchangeably (Fig. 7), as it was also found by Kattenborn et al. (2019) for pigments. One essential reason for the preferred usage of the mass-based measures might be that leaf nutrients are commonly extracted from plant powder without the availability of either the covered area of leaves or ground (Kattenborn et al., 2019). Therefore, field experiments measuring both crop biochemistry as well as structural canopy variables are highly encouraged.

Some studies focused on the estimation of N uptake although this goal might be challenging to be achieved solely by remote sensing. Hereby, the time factor (i.e. crop growth rate) must be taken into account including the N allocation between vegetative and mature or reproductive organs. For instance, in case of paddy rice approximately 80% of total plant N is absorbed from the soil before flowering and stored in vegetative organs. At the same time and subsequently, N is transported from these organs to the ears and the grains (Ohyama, 2010). There is a strong link between cumulative N uptake and aboveground biomass accumulation. A mathematical relationship between these parameters has been introduced by Plenet and Lemaire (1999), including the amount of N present in the shoots as well as the actual aboveground biomass.

Effectively, the aboveground N_{area} could be estimated for a certain point of time by means of remote sensing techniques. The determination of N uptake, instead, is often not straightforward. This could be potentially achieved with the knowledge of N allocation as well as by combining remote sensing with process-based crop growth models.

Finally, it should be pointed out that remote sensing is primarily limited to N retrieval in leaves. It is not possible to directly detect N stored in the soil. The same holds true for plant organs such as tubers, fruits or hardened stalks because radiation in the solar optical domain is not able to penetrate these tissues completely and to transport information about the biochemical content. Accordingly, a reference to "aboveground nitrogen content" seems most appropriate to describe N content at the canopy scale. Nevertheless, we recommend adapting terminology and units according to the related objectives and hypotheses of the specific studies.

6.5 Study locations and analyzed crop types

The largest part of the studies was performed in China (~45% of all studies), which can be easily explained by the large agricultural areas and the increasing usage of remote sensing techniques and other precision farming technology. European countries follow with 27%, with most studies performed in Germany and Italy. The USA with their vast agricultural regions holds 15% of the studies. Studies were carried out in all continents except South America. This large number of countries involved reflects the global interest in optimal N supply for agriculture.

Regarding the crop types, wheat or winter wheat was analyzed with 34% of all studies, followed by rice with 21% and maize with 13%. This mirrors the world's staple cropping systems with wheat, rice and maize occupying a large part of agricultural areas around the globe (Khoury et al., 2014).

7 Conclusions

In recent years, interest grew in the determination of crop N using imaging spectroscopy (hyperspectral) data (Fig. 3). This need is currently addressed by the launch and planning of several satellite imaging spectrometers - suitable for N sensing due to the included SWIR domain - that will provide scenes over large agricultural areas. High spatial and temporal coverages supplied by these stable instruments optimally designed for the retrieval of nitrogen will allow testing the transferability of all mentioned N retrieval approaches.

From our meta-analysis, we identified some specific problems and limitations: at first, main effort was invested into parametric regression modelling despite the well-known limitations of vegetation indices, mostly not discussed in the studies. The increasing usage of chemometrics with the most popularly applied PLSR method leads to improved retrievals. Nonetheless, chemometrics still suffer from limited transferability and the inevitable need of in situ training data. Among the machine learning algorithms, GPRs were rarely used, although this technique can be considered as one of the most promising regression models providing uncertainties of the estimates. Within all methods, a strong focus on the N-chlorophyll relationship could be identified, neglecting the fact that most N is bound in proteins. Moreover, there is a lot of confusion in used terminology, i.e. N content versus N concentration, or many other terms. We therefore recommend a standardization of terms, using mass-based N (N concentration, N%) and area-based N (N content, N_{area}). In general, a dominance of N%-based studies compared to N_{area}-based studies can be observed, which may be related to the common measurement practices. The apparent estimation of "N

uptake" from hyperspectral signatures should be handled carefully, because the processes of N allocation between vegetative and reproductive organs and the strong connection to the aboveground biomass accumulation must be considered. However, if there is ground data of biomass available (for instance using LiDAR measurements or biophysical crop modelling) then the term "N uptake" can be appropriate. In all cases, the authors should always provide the appropriate units of their studied measures.

We further recommend the development of leaf RTMs simulating absorption features of proteins. In this way, the retrieval of N solely based on chlorophyll content could be overcome. Nevertheless, one of the main reasons for focusing on chlorophyll measurements and estimations is to understand photosynthetic processes, plant stress and impact to plant growth and development, as well as N uptake. Therefore, decoupling of the estimation of chlorophyll and N content would provide a more pronounced understanding of N allocation within the plant and the origin of diverse stresses. Moreover, the application of hybrid methods, in particular the combination of physically-based retrieval combined with different machine learning algorithms is suggested for N sensing. In this way, a large training dataset can be generated from RTM simulations which reduces the need for collecting in situ training data. Nevertheless, successful validation exercises under various conditions are required to allow for transfer of the proposed hybrid inversion approach to agricultural areas around the globe: therefore, we strongly recommend to organize extensive field campaigns during future hyperspectral satellite sensors overpasses with simultaneous structural (biomass, LAI) and biochemical (chlorophyll, water, nitrogen, LMA) sampling for improved crop N determination from hyperspectral data. Concurrently or independently, proximal sensing studies (using field spectrometers) can help developing the tools and methods to be further applied and validated at the canopy scale using satellite data.

Continuous crop N monitoring may help to understand the transition of N from the soil into the crop, to help farmers to improve their farm management (Hank et al., 2019) and to alert growers about potential shortfalls, thus guiding real-time or long-term fertilizer management plan modifications for future production years. In this way, overfertilization could be reduced preventing harmful effects on the environment, while at the same time a sustainable use of production resources can be maintained. The outcomes of our study propose that a new era of N sensing also beyond agricultural research and applications could be initiated.

Supplementary data to this article can be found online at https://doi.org/10.1016/j.rse.2020.111758.

Supplementary Material

Refer to Web version on PubMed Central for supplementary material.

Acknowledgements

The results shown in this paper were achieved as part of the research project "EnMAP Scientific Advisory Group Phase III Developing the EnMAP Managed Vegetation Scientific Processor", supported by the Space Agency of the German Aerospace Center (DLR) through funding by the German Federal Ministry for Economic Affairs and Energy under the grant code 50EE1623. Jochem Verrelst was supported by the European Research Council

(ERC) under the ERC-2017-STG SENTIFLEX project (grant agreement 755617) and Ramón y Cajal Contract (Spanish Ministry of Science, Innovation and Universities). J.-B. Féret acknowledges financial support from Agence Nationale de la Recherche (BioCop project—ANR-17-CE32-0001) and TOSCA program grant of the French Space Agency (CNES) (HyperTropik/HyperBIO project). We thank the three reviewers for their valuable suggestions.

References

- Abdel-Rahman EM, Ahmed FB, Ismail R. Random forest regression and spectral band selection for estimating sugarcane leaf nitrogen concentration using EO-1 Hyperion hyperspectral data. Int J Remote Sens. 2013; 34: 712–728.
- A A, Malenovský Z, Olejní ková J, Gallé A, Rascher U, Mohammed G. Meta-analysis assessing potential of steady-state chlorophyll fluorescence for remote sensing detection of plant water, temperature and nitrogen stress. Remote Sens Environ. 2015; 168: 420–436.
- Aber, J, editor. ACCP. Accelerated Canopy Chemistry Program Final Report to NASA-EOS-IWG. Washington, DC: 1994.
- Afandi SD, Herdiyeni Y, Prasetyo LB, Hasbi W, Arai K, Okumura H. Nitrogen content estimation of rice crop based on near infrared (NIR) reflectance using artificial neural network (ANN). Procedia Environ Sci. 2016; 33: 63–69.
- Alchanatis V, Schmilovitch Z, Meron M. In-field assessment of single leaf nitrogen status by spectral reflectance measurements. Precis Agric. 2005; 6: 25–39.
- Ashourloo D, Aghighi H, Matkan AA, Mobasheri MR, Rad AM. An investigation into machine learning regression techniques for the leaf rust disease detection using hyperspectral measurement. IEEE J Sel Top Appl Earth Obs Remote Sens. 2016; 9: 4344–4351.
- Asner G, Martin R. Airborne spectranomics: mapping canopy chemical and taxonomic diversity in tropical forests. Front Ecol Environ. 2009; 7: 269–276.
- Atzberger C. Advances in remote sensing of agriculture: context description, existing operational monitoring systems and major information needs. Remote Sens. 2013; 5: 949–981.
- Atzberger C, Richter K. Spatially constrained inversion of radiative transfer models for improved LAI mapping from future Sentinel-2 imagery. Remote Sens Environ. 2012; 120: 208–218.
- Atzberger C, Guérif M, Baret F, Werner W. Comparative analysis of three chemometric techniques for the spectroradiometric assessment of canopy chlorophyll content in winter wheat. Comput Electron Agric. 2010; 73: 165–173.
- Atzberger, C, Berger, K, Vuolo, F, Darvishzadeh, R, Schlerf, M. Why confining to vegetation indices? Exploiting the potential of improved spectral observations using radiative transfer models. SPIE Remote Sensing Europe; Prague, Czech Republic: 2011.
- Bach, H; Mauser, W; Gernot, K. In: Ouwehand, L, editor. Earth observation for food security and sustainable agriculture; Living Planet Symposium; Prague, Czech Republic. 2016. 96
- Baker RE, Pena JM, Jayamohan J, Jerusalem A. Mechanistic models versus machine learning, a fight worth fighting for the biological community? Biol Lett. 2018; 14
- Baret, F, Buis, S. Advances in Land Remote Sensing. Springer; Dordrecht, Netherlands: 2008. 173– 201.
- Baret, F, Fourty, TH, L, G. Diagnosis of the Nitrogen Status in Crops. Springer; Berlin, Heidelberg: 1997. 201–227.
- Baret F, Houles V, Guerif M. Quantification of plant stress using remote sensing observations and crop models: the case of nitrogen management. J Exp Bot. 2007; 58: 869–880. [PubMed: 17220515]
- Bassham JA, Benson AA, Kay LD, Harris AZ, Wilson AT, Calvin M. The path of carbon in photosynthesis. XXI. The cyclic regeneration of carbon dioxide acceptor1. J Am Chem Soc. 1954; 76: 1760–1770.
- Beeckman F, Motte H, Beeckman T. Nitrification in agricultural soils: impact, actors and mitigation. Curr Opin Biotechnol. 2018; 50: 166–173. [PubMed: 29414056]
- Belgiu M, Dr gu L. Random forest in remote sensing: a review of applications and future directions. Isprs J Photogramm. 2016; 114: 24–31.

- Berger K, Atzberger C, Danner M, D'Urso G, Mauser W, Vuolo F, Hank T. Evaluation of the PROSAIL model capabilities for future hyperspectral model environments: a review study. Remote Sens. 2018a; 10: 85.
- Berger, K; Wang, Z; Danner, M; Wocher, M; Mauser, W; Hank, T. Simulation of spaceborne hyperspectral remote sensing to assist crop nitrogen content monitoring in agricultural crops; IEEE International Geoscience and Remote Sensing Symposium; Valencia, Spain. 2018b. 3801–3804.
- Boegh E, Soegaard H, Broge N, Hasager CB, Jensen NO, Schelde K, Thomsen A. Airborne multispectral data for quantifying leaf area index, nitrogen concentration, and photosynthetic efficiency in agriculture. Remote Sens Environ. 2002; 81: 179–193.

Bravo C, Moshou D, West J, McCartney A, Ramon H. Early disease detection in wheat fields using spectral reflectance. Biosyst Eng. 2003; 84: 137–145.

- Breiman L. Random forests. Mach Learn. 2001; 45: 5-32.
- Camino C, González-Dugo V, Hernández P, Sillero JC, Zarco-Tejada PJ. Improved nitrogen retrievals with airborne-derived fluorescence and plant traits quantified from VNIR-SWIR hyperspectral imagery in the context of precision agriculture. Int J Appl Earth Obs Geoinf. 2018; 70: 105–117.
- Camps-Valls G, Verrelst J, Munoz-Mari J, Laparra V, Mateo-Jimenez F, Gomez-Dans J. A survey on Gaussian processes for earth-observation data analysis: a comprehensive investigation. IEEE Geosc Rem Sen M. 2016; 4: 58–78.
- Camps-Valls G, Sejdinovic D, Runge J, Reichstein M. A perspective on Gaussian processes for earth observation. Natl Sci Rev. 2019; 6: 616–618. [PubMed: 34691913]
- Castaldi F, Castrignano A, Casa R. A data fusion and spatial data analysis approach for the estimation of wheat grain nitrogen uptake from satellite data. Int J Remote Sens. 2016; 37: 4317–4336.
- Chapin FS, Bloom AJ, Field CB, Waring RH. Plant responses to multiple environmental factors. Bioscience. 1987; 37: 49–57.
- Chen P, Haboudane D, Tremblay N, Wang J, Vigneault P, Li B. New spectral indicator assessing the efficiency of crop nitrogen treatment in corn and wheat. Remote Sens Environ. 2010; 114: 1987–1997.
- Chen J, Li F, Wang R, Fan Y, Raza MA, Liu Q, Wang Z, Cheng Y, Wu X, Yang F, Yang W. Estimation of nitrogen and carbon content from soybean leaf reflectance spectra using wavelet analysis under shade stress. Comput Electron Agric. 2019; 156: 482–489.
- Chlingaryan A, Sukkarieh S, Whelan B. Machine learning approaches for crop yield prediction and nitrogen status estimation in precision agriculture: a review. Comput Electron Agric. 2018; 151: 61–69.
- Cidad, VG, Vrindts, E, De Baerdemaeker, J. Europto Remote Sensing, Remote Sensing for Agriculture, Ecosystems, and Hydrology II. Owe, M, D'Urso, G, Zilioli, E, editors. SPIE; Barcelona, Spain: 2000.
- Clevers J, Kooistra L. Using hyperspectral remote sensing data for retrieving canopy chlorophyll and nitrogen content. IEEE J Sel Top Appl Earth Obs Remote Sens. 2012; 5: 574–583.
- Curran PJ. Remote sensing of foliar chemistry. Remote Sens Environ. 1989; 30: 271-278.
- Dale L, Thewis A, Boudry C, Rotar I, Dardenne P, Baeten V, Fernández Pierna J. Hyperspectral imaging applications in agriculture and agro-food product quality and safety control: a review. Appl Spectrosc Rev. 2013; 48: 142.
- Danner M, Berger K, Wocher M, Mauser W, Hank T. Retrieval of biophysical crop variables from multi-angular canopy spectroscopy. Remote Sens. 2017; 9: 726.
- Darvishzadeh R, Wang T, Skidmore A, Vrieling A, O'Connor B, Gara TW, Ens BJ, Paganini M. Analysis of Sentinel-2 and RapidEye for retrieval of leaf area index in a saltmarsh using a radiative transfer model. Remote Sens. 2019; 11: 671.
- Davies AMC. What is and what is not chemometrics. Spectrosc Eur. 2012; 24: 33-36.
- Dechant B, Cuntz M, Vohland M, Schulz E, Doktor D. Estimation of photosynthesis traits from leaf reflectance spectra: correlation to nitrogen content as the dominant mechanism. Remote Sens Environ. 2017; 196: 279–292.
- Dise, NB, Ashmore, M, Belyazid, S, Bleeker, A, Bobbink, R, de Vries, W, Erisman, JW, Spranger, T, Stevens, CJ, van den Berg, L. The European Nitrogen Assessment: Sources, Effects and Policy

Perspectives. Bleeker, A, Grizzetti, B, Howard, CM, Billen, G, van Grinsven, H, Erisman, JW, Sutton, MA, Grennfelt, P, editors. Cambridge University Press; Cambridge: 2011. 463–494.

- Doktor D, Lausch A, Spengler D, Thurner M. Extraction of plant physiological status from hyperspectral signatures using machine learning methods. Remote Sens. 2014; 6: 12247–12274.
- Dunn BW, Dehaan R, Schmidtke LM, Dunn TS, Meder R. Using Field-derived hyperspectral reflectance measurement to identify the essential wavelengths for predicting nitrogen uptake of Rice at panicle initiation. J Near Infrared Spectrosc. 2016; 24: 473–483.
- Ecarnot M, Compan F, Roumet P. Assessing leaf nitrogen content and leaf mass per unit area of wheat in the field throughout plant cycle with a portable spectrometer. Field Crops Res. 2013; 140: 44–50.
- Eitel JUH, Long DS, Gessler PE, Hunt ER. Combined spectral index to improve ground-based estimates of nitrogen status in dryland wheat. Agron J. 2008; 100: 1694–1702.
- Ellis RJ. The most abundant protein in the world. Trends Biochem Sci. 1979; 4: 241-244.
- Elvidge CD. Visible and near infrared reflectance characteristics of dry plant materials. Int J Remote Sens. 1990; 11: 1775–1795.
- Evans JR. Photosynthesis and nitrogen relationships in leaves of C3 plants. Oecologia. 1989; 78: 9–19. [PubMed: 28311896]
- Feingersh, T, Ben Dor, E. Optical Payloads for Space Missions. Qian, S, editor. John Wiley & Sons, Ltd; 2015. 247–263.
- Feng W, Guo B-B, Wang Z-J, He L, Song X, Wang Y-H, Guo T-C. Measuring leaf nitrogen concentration in winter wheat using double-peak spectral reflection remote sensing data. Field Crops Res. 2014; 159: 43–52.
- Feret J-B, François C, Asner GP, Gitelson AA, Martin RE, Bidel LPR, Ustin SL, le Maire G, Jacquemoud S. PROSPECT-4 and 5: advances in the leaf optical properties model separating photosynthetic pigments. Remote Sens Environ. 2008; 112: 3030–3043.
- Féret JB, Gitelson AA, Noble SD, Jacquemoud S. PROSPECT-D: towards modeling leaf optical properties through a complete lifecycle. Remote Sens Environ. 2017; 193: 204–215.
- Féret JB, le Maire G, Jay S, Berveiller D, Bendoula R, Hmimina G, Cheraiet A, Oliveira JC, Ponzoni FJ, Solanki T, de Boissieu F, et al. Estimating leaf mass per area and equivalent water thickness based on leaf optical properties: Potential and limitations of physical modeling and machine learning. Remote Sens Environ. 2019; 231 110959
- Ferwerda JG, Skidmore AK, Mutanga O. Nitrogen detection with hyperspectral normalized ratio indices across multiple plant species. Int J Remote Sens. 2005; 26: 4083–4095.
- Filella I, Serrano L, Serra J, Peñuelas J. Evaluating wheat nitrogen status with canopy reflectance indices and discriminant analysis. Crop Sci. 1995; 35: 1400–1405.
- Fourty T, Baret F. Vegetation water and dry matter contents estimated from top-of-the-atmosphere reflectance data: a simulation study. Remote Sens Environ. 1997; 61: 34–45.
- Fourty T, Baret F, Jacquemoud S, Schmuck G, Verdebout J. Leaf optical properties with explicit description of its biochemical composition: direct and inverse problems. Remote Sens Environ. 1996; 56: 104–117.
- Gao, F, Morisette, J, Wolfe, R, Ederer, G, Pedelty, J, Ed, M, Myneni, R, Tan, B. Geoscience and Remote Sensing Letters. Vol. 5. IEEE; 2008. 60–64.
- Gao B-C, Montes MJ, Davis CO, Goetz AFH. Atmospheric correction algorithms for hyperspectral remote sensing data of land and ocean. Remote Sens Environ. 2009; 113: S17–S24.
- Gillis, J. Web of Science. Clarivate Analytics; 22 Thomson Place, 36T3 Boston, MA 02210: 2018.
- Goel PK, Prasher SO, Landry JA, Patel RM, Bonnell RB, Viau AA, Miller JR. Potential of airborne hyperspectral remote sensing to detect nitrogen deficiency and weed infestation in corn. Comput Electron Agric. 2003; 38: 99–124.
- Goffart JP, Olivier M, Frankinet M. Potato crop nitrogen status assessment to improve N fertilization management and efficiency: past-present-future. Potato Res. 2008; 51: 355–383.
- Greenwood DJ, Neeteson JJ, Draycott A. Quantitative relationships for the dependence of growth rate of arable crops on their nitrogen content, dry weight and aerial environment. Plant Soil. 1986; 91: 281–301.

- Grossman YL, Ustin SL, Jacquemoud S, Sanderson EW, Schmuck G, Verdebout J. Critique of stepwise multiple linear regression for the extraction of leaf biochemistry information from leaf reflectance data. Remote Sens Environ. 1996; 56: 182–193.
- Guanter L, Kaufmann H, Segl K, Foerster S, Rogass C, Chabrillat S, Kuester T, Hollstein A, Rossner G, Chlebek C, Straif C, et al. The EnMAP spaceborne imaging spectroscopy mission for earth observation. Remote Sens. 2015; 7: 8830–8857.
- Hallik L, Kull O, Niinemets Ü, Aan A. Contrasting correlation networks between leaf structure, nitrogen and chlorophyll in herbaceous and woody canopies. Basic Appl Ecol. 2009; 10: 309–318.
- Hank TB, Berger K, Bach H, Clevers JGPW, Gitelson A, Zarco-Tejada P, Mauser W. Spaceborne imaging spectroscopy for sustainable agriculture: contributions and challenges. Surv Geophys. 2019; 40: 515–551.
- Hansen PM, Schjoerring JK. Reflectance measurement of canopy biomass and nitrogen status in wheat crops using normalized difference vegetation indices and partial least squares regression. Remote Sens Environ. 2003; 86: 542–553.
- He L, Song X, Feng W, Guo B-B, Zhang Y-S, Wang Y-H, Wang C-Y, Guo T-C. Improved remote sensing of leaf nitrogen concentration in winter wheat using multi-angular hyperspectral data. Remote Sens Environ. 2016; 174: 122–133.
- Herrmann I, Karnieli A, Bonfil DJ, Cohen Y, Alchanatis V. SWIR-based spectral indices for assessing nitrogen content in potato fields. Int J Remote Sens. 2010; 31: 5127–5143.
- Himmelsbach, DS, Boer, H, Akin, DE, Barton, FEI. Analytical Applications of Spectroscopy. Davies, AMC, editor. Royal Society of Chemistry; London: 1988.
- Homolová L, Malenovský Z, Clevers JGPW, García-Santos G, Schaepman ME. Review of opticalbased remote sensing for plant trait mapping. Ecol Complex. 2013; 15: 1–16.
- Horler DNH, Dockray M, Barber J. The red edge of plant leaf reflectance. Int J Remote Sens. 1983; 4: 273–288.
- Hosgood, B, Jacquemoud, S, Andreoli, G, Verdebout, J, Pedrini, G, Schmuck, G. Leaf Optical Properties Experiment 93 (LOPEX93). I.f.R.S.A. European Commission-Joint Research Centre (Ed.), (EUR 16095 EN); 1994.
- Hunt ER, Doraiswamy PC, McMurtrey JE, Daughtry CST, Perry EM, Akhmedov B. A visible band index for remote sensing leaf chlorophyll content at the canopy scale. Int J Appl Earth Obs Geoinf. 2013; 21: 103–112.
- Inoue Y, Sakaiya E, Zhu Y, Takahashi W. Diagnostic mapping of canopy nitrogen content in rice based on hyperspectral measurements. Remote Sens Environ. 2012; 126: 210–221.
- Inoue Y, Darvishzadeh R, Skidmore A. Hyperspectral assessment of ecophysiological functioning for diagnostics of crops and vegetation. Thenkabail PS, Lyon JG, Huete AR. Biophysical and Biochemical Characterization and Plant Species Studies. 2018. 25–71.
- Jacquemoud S, Baret F. PROSPECT: a model of leaf optical properties spectra. Remote Sens Environ. 1990; 34: 75–91.
- Jacquemoud S, Verdebout J, Schmuck G, Andreoli G, Hosgood B. Investigation of leaf biochemistry by statistics. Remote Sens Environ. 1995; 54: 180–188.
- Jacquemoud S, Ustin SL, Verdebout J, Schmuck G, Andreoli G, Hosgood B. Estimating leaf biochemistry using the PROSPECT leaf optical properties model. Remote Sens Environ. 1996; 56: 194–202.
- Jacquemoud S, Verhoef W, Baret F, Bacour C, Zarco-Tejada PJ, Asner GP, François C, Ustin SL. PROSPECT + SAIL models: a review of use for vegetation characterization. Remote Sens Environ. 2009; 113 (Supplement 1) S56–S66.
- Jay S, Maupas F, Bendoula R, Gorretta N. Retrieving LAI, chlorophyll and nitrogen contents in sugar beet crops from multi-angular optical remote sensing: comparison of vegetation indices and PROSAIL inversion for field phenotyping. Field Crops Res. 2017; 210: 33–46.
- Jaynes DB, Colvin TS, Karlen DL, Cambardella CA, Meek DW. Nitrate loss in subsurface drainage as affected by nitrogen fertilizer rate. J Environ Qual. 2001; 30: 1305–1314. [PubMed: 11476509]
- Jia M, Zhu J, Ma C, Alonso L, Li D, Cheng T, Tian Y, Zhu Y, Yao X, Cao W. Difference and potential of the upward and downward sun-induced chlorophyll fluorescence on detecting leaf nitrogen concentration in wheat. Remote Sens. 2018; 10: 1315.

- Justes E, Mary B, Meynard JM, Machet JM, Thelier-Huche L. Determination of a critical nitrogen dilution curve for winter wheat crops. Ann Bot. 1994; 74: 397–407.
- Kattenborn T, Schiefer F, Zarco-Tejada P, Schmidtlein S. Advantages of retrieving pigment content [µg/cm²] versus concentration [%] from canopy reflectance. Remote Sens Environ. 2019; 230 111195
- Kattge, J. Institut für Pflanzenökologie. Justus-Liebig-Universität; Gießen: 2002. 285
- Kattge J, Knorr W, Raddatz T, Wirth C. Quantifying photosynthetic capacity and its relationship to leaf nitrogen content for global-scale terrestrial biosphere models. Glob Chang Biol. 2009; 15: 976–991.
- Keshava N, Mustard JF. Spectral unmixing. IEEE Signal Process Mag. 2002; 19: 44-57.
- Khoury CK, Bjorkman AD, Dempewolf H, Ramirez-Villegas J, Guarino L, Jarvis A, Rieseberg LH, Struik PC. Increasing homogeneity in global food supplies and the implications for food security. Proc Natl Acad Sci. 2014; 111: 4001–4006. [PubMed: 24591623]
- Kimes DS, Knyazikhin Y, Privette JL, Abuelgasim AA, Gao F. Inversion methods for physically-based models. Remote Sens Rev. 2000; 18: 381–439.
- Kimura, E, Bell, J, Trostle, C, Neely, C, Drake, D. Potential Causes of Yellowing During the Tillering Stage of Wheat in Texas. Texas A&M AgriLife Extension ESC-040; 2016.
- Knyazikhin Y, Lewis P, Disney M, Stenberg P, Mõttus M, Rautiainen M, Kaufmann R, Marshak A, Schull M, Carmona P, Vanderbilt V, et al. Reply to Townsend et al.: decoupling contributions from canopy structure and leaf optics is critical for remote sensing leaf biochemistry. Proc Natl Acad Sci U S A. 2013a; 110 E1075 [PubMed: 23630977]
- Knyazikhin Y, Schull MA, Stenberg P, Mottus M, Rautiainen M, Yang Y, Marshak A, Latorre Carmona P, Kaufmann RK, Lewis P, Disney MI, et al. Hyperspectral remote sensing of foliar nitrogen content. Proc Natl Acad Sci U S A. 2013b; 110: E185–E192. [PubMed: 23213258]
- Kokaly RF, Clark RN. Spectroscopic determination of leaf biochemistry using band-depth analysis of absorption features and stepwise multiple linear regression. Remote Sens Environ. 1999; 67: 267–287.
- Kokaly RF, Asner GP, Ollinger SV, Martin ME, Wessman CA. Characterizing canopy biochemistry from imaging spectroscopy and its application to ecosystem studies. Remote Sens Environ. 2009; 113: S78–S91.
- Kumar, L, Schmidt, K, Dury, S, Skidmore, A. Imaging Spectrometry: Basic Principles and Prospective Applications. Meer, FDvd; Jong, SMD, editors. Springer; Netherlands, Dordrecht: 2001. 111–155.
- Lavine BK, Workman J. Chemometrics. Anal Chem. 2013; 85: 705–714. [PubMed: 23140170]
- Lee Y-J, Yang C-M, Chang K-W, Shen Y. A simple spectral index using reflectance of 735 nm to assess nitrogen status of rice canopy. Agron J. 2008; 100: 205–212.
- Lee C, Cable M, Hook S, Green R, Ustin S, Mandl D, Middleton E. An introduction to the NASA Hyperspectral InfraRed Imager (HyspIRI) mission and preparatory activities. Remote Sens Environ. 2015; 167: 6–19.
- Lelong CCD, Pinet PC, Poilvé H. Hyperspectral imaging and stress mapping in agriculture: a case study on wheat in Beauce (France). Remote Sens Environ. 1998; 66: 179–191.
- Lemaire, G, Gastal, F, Salette, J. Analysis of the effect of N nutrition on dry matter yield of a sward by reference to potential yield and optimum N content. Vol. 16. International Grassland Congress; 1989. 179–180. 1989-10-04
- Lemaire G, Jeuffroy M-H, Gastal F. Diagnosis tool for plant and crop N status in vegetative stage: theory and practices for crop N management. Eur J Agron. 2008; 28: 614–624.
- Li F, Miao YX, Hennig SD, Gnyp ML, Chen XP, Jia LL, Bareth G. Evaluating hyperspectral vegetation indices for estimating nitrogen concentration of winter wheat at different growth stages. Precis Agric. 2010; 11: 335–357.
- Li D, Wang X, Zheng HB, Zhou K, Yao X, Tian YC, Zhu Y, Cao WX, Cheng T. Estimation of areaand mass-based leaf nitrogen contents of wheat and rice crops from water-removed spectra using continuous wavelet analysis. Plant Methods. 2018; 14: 20. [PubMed: 29563961]
- Li L, Jákli B, Lu P, Ren T, Ming J, Liu S, Wang S, Lu J. Assessing leaf nitrogen concentration of winter oilseed rape with canopy hyperspectral technique considering a non-uniform vertical nitrogen distribution. Ind Crop Prod. 2018; 116: 1–14.

- Li Z, Jin X, Yang G, Drummond J, Yang H, Clark B, Li Z, Zhao C. Remote sensing of leaf and canopy nitrogen status in winter wheat (Triticum aestivum L.) based on N-PROSAIL model. Remote Sens. 2018; 10: 1463.
- Li Z, Li Z, Fairbairn D, Li N, Xu B, Feng H, Yang G. Multi-LUTs method for canopy nitrogen density estimation in winter wheat by field and UAV hyperspectral. Comput Electron Agric. 2019; 162: 174–182.
- Liang L, Di L, Huang T, Wang J, Lin L, Wang L, Yang M. Estimation of leaf nitrogen content in wheat using new hyperspectral indices and a random forest regression algorithm. Remote Sens. 2018; 10: 1940.
- Liu L, Song B, Zhang S, Liu X. A novel principal component analysis method for the reconstruction of leaf reflectance spectra and retrieval of leaf biochemical contents. Remote Sens. 2017; 9: 1113.
- Loizzo, R; Daraio, M; Guarini, R; Longo, F; Lorusso, R; Dini, L; Lopinto, E. Prisma Mission status and perspective; IEEE International Geoscience and Remote Sensing Symposium (IGARSS); Yokohama, Japan. 2019. 4503–4506.
- Makino A. Rubisco and nitrogen relationships in rice: leaf photosynthesis and plant growth. Soil Sci Plant Nutr. 2003; 49: 319–327.
- Malenovský Z, Homolová L, Zurita-Milla R, Lukeš P, Kaplan V, Hanuš J, Gastellu-Etchegorry J-P, Schaepman ME. Retrieval of spruce leaf chlorophyll content from airborne image data using continuum removal and radiative transfer. Remote Sens Environ. 2013; 131: 85–102.
- Malenovský Z, Homolová L, Lukeš P, Buddenbaum H, Verrelst J, Alonso L, Schaepman ME, Lauret N, Gastellu-Etchegorry J-P. Variability and uncertainty challenges in scaling imaging spectroscopy retrievals and validations from leaves up to vegetation canopies. Surv Geophys. 2019; 40: 631–656.
- Mariotti F, Tomé D, Mirand P. Converting Nitrogen into Protein—Beyond 6.25 and Jones' Factors. Crit Rev Food Sci Nutr. 2008; 48: 177–184. [PubMed: 18274971]
- Marshall M, Thenkabail P. Advantage of hyperspectral EO-1 Hyperion over multispectral IKONOS, GeoEye-1, WorldView-2, Landsat ETM+, and MODIS vegetation indices in crop biomass estimation. Isprs J Photogramm. 2015; 108: 205–218.
- Marshall M, Thenkabail P, Biggs T, Post K. Hyperspectral narrowband and multispectral broadband indices for remote sensing of crop evapotranspiration and its components (transpiration and soil evaporation). Agric For Meteorol. 2016; 218–219: 122–134.
- Masclaux-Daubresse C, Daniel-Vedele F, Dechorgnat J, Chardon F, Gaufichon L, Suzuki A. Nitrogen uptake, assimilation and remobilization in plants: challenges for sustainable and productive agriculture. Ann Bot. 2010; 105: 1141–1157. [PubMed: 20299346]
- Matsunaga, T; Iwasaki, A; Tsuchida, S; Iwao, K; Tanii, J; Kashimura, O; Nakamura, R; Yamamoto, H; Kato, S; Obata, K; Mouri, K; , et al. Current status of Hyperspectral Imager Suite (HISUI) onboard International Space Station (ISS); 2017 IEEE International Geoscience and Remote Sensing Symposium (IGARSS); 2017. 443–446.
- Milford GFJ, Pocock TO, Riley J, Messem AB. An analysis of leaf growth in sugar beet. III. Leaf expansion in field crops. Ann Appl Biol. 1985; 106: 187–203.
- Milton K, Dintzis FR. Nitrogen-to-protein conversion factors for tropical plant samples. Biotropica. 1981; 13: 177–181.
- Miphokasap P, Wannasiri W. Estimations of nitrogen concentration in sugarcane using hyperspectral imagery. Sustainability. 2018; 10: 1266.
- Miphokasap P, Honda K, Vaiphasa C, Souris M, Nagai M. Estimating canopy nitrogen concentration in sugarcane using Field imaging spectroscopy. Remote Sens. 2012; 4: 1651.
- Mohammed GH, Colombo R, Middleton EM, Rascher U, van der Tol C, Nedbal L, Goulas Y, Pérez-Priego O, Damm A, Meroni M, Joiner J, et al. Remote sensing of solar-induced chlorophyll fluorescence (SIF) in vegetation: 50 years of progress. Remote Sens Environ. 2019; 231 111177 [PubMed: 33414568]
- Moharana S, Dutta S. Spatial variability of chlorophyll and nitrogen content of rice from hyperspectral imagery. Isprs J Photogramm. 2016; 122: 17–29.
- Monteiro ST, Minekawa Y, Kosugi Y, Akazawa T, Oda K. Prediction of sweetness and amino acid content in soybean crops from hyperspectral imagery. Isprs J Photogramm. 2007; 62: 2–12.

- Mosse J. Nitrogen-to-protein conversion factor for ten cereals and six legumes or oilseeds. A reappraisal of its definition and determination. Variation according to species and to seed protein content. J Agric Food Chem. 1990; 38: 18–24.
- Mulla DJ. Twenty five years of remote sensing in precision agriculture: key advances and remaining knowledge gaps. Biosyst Eng. 2013; 114: 358–371.
- Muñoz-Huerta R, Guevara-Gonzalez R, Contreras-Medina L, Torres-Pacheco I, Prado-Olivarez J, Ocampo-Velazquez R. A review of methods for sensing the nitrogen status in plants: advantages, disadvantages and recent advances. Sensors. 2013; 13 10823 [PubMed: 23959242]
- Mutanga O, Adam E, Adjorlolo C, Abdel-Rahman EM. Evaluating the robustness of models developed from field spectral data in predicting African grass foliar nitrogen concentration using WorldView-2 image as an independent test dataset. Int J Appl Earth Obs Geoinf. 2015; 34: 178–187.
- NASA. HyspIRI Mission Concept Team. Jet Propulsion Laboratory, California Institute of Technology; Pasadena, California: 2018.
- Näsi R, Viljanen N, Kaivosoja J, Alhonoja K, Hakala T, Markelin L, Honkavaara E. Estimating biomass and nitrogen amount of barley and grass using UAV and aircraft based spectral and photogrammetric 3D features. Remote Sens. 2018; 10: 1082.
- Natale, V; Kafri, A; Tidhar, G; Chen, M; Feingersh, T; Sagi, E; Cisbani, A; Baroni, M; Labate, D; Nadler, R; Leizer, A; , et al. SHALOM—Space-borne hyperspectral applicative land and ocean mission; 5th Workshop on Hyperspectral Image and Signal Processing: Evolution in Remote Sensing (WHISPERS); Gainesville, FL, USA. 2013. 1–4.
- Nieke, J; Rast, M. Towards the Copernicus Hyperspectral Imaging Mission For The Environment (CHIME); IGARSS 2018-2018 IEEE International Geoscience and Remote Sensing Symposium; Valencia, Spain. 2018. 157–159.
- Nigon TJ, Mulla DJ, Rosen CJ, Cohen Y, Alchanatis V, Knight J, Rud R. Hyperspectral aerial imagery for detecting nitrogen stress in two potato cultivars. Comput Electron Agric. 2015; 112: 36–46.
- Nørgaard L, Saudland A, Wagner J, Nielsen JP, Munck L, Engelsen SB. Interval partial least-squares regression (iPLS): a comparative chemometric study with an example from near-infrared spectroscopy. Appl Spectrosc. 2000; 54: 413–419.
- Ohyama, T. Nitrogen Assimilation in Plants. Ohyama, Takuji; Sueyoshi, K, editors. Research Signpost; 2010. 1–18.
- Ollinger S. Sources of variability in canopy reflectance and the convergent properties of plants. The New phytologist. 2010; 189: 375–394. [PubMed: 21083563]
- Osnas JLD, Lichstein JW, Reich PB, Pacala SW. Global leaf trait relationships: mass, area, and the leaf economics spectrum. Science. 2013; 340: 741–744. [PubMed: 23539179]
- Padilla FM, Gallardo M, Pena-Fleitas MT, de Souza R, Thompson RB. Proximal optical sensors for nitrogen management of vegetable crops: a review. Sensors (Basel). 2018; 18
- Pellissier PA, Ollinger SV, Lepine LC, Palace MW, McDowell WH. Remote sensing of foliar nitrogen in cultivated grasslands of human dominated landscapes. Remote Sens Environ. 2015; 167: 88– 97.
- Peñuelas J, Gamon JA, Fredeen AL, Merino J, Field CB. Reflectance indices associated with physiological changes in nitrogen- and water-limited sunflower leaves. Remote Sens Environ. 1994; 48: 135–146.
- Perry EM, Roberts DA. Sensitivity of narrow-band and broad-band indices for assessing nitrogen availability and water stress in an annual crop. Agron J. 2008; 100: 1211–1219.
- Plenet D, Lemaire G. Relationships between dynamics of nitrogen uptake and dry matter accumulation in maize crops. Determination of critical N concentration. Plant Soil. 1999; 216: 65–82.
- Powell, CW, Lindquist, RK. Ball Pest and Disease Manual: Disease, Insect, and Mite Control on Flower and Foliage Crops. Ball Publishing; Batavia, Illinois: 1997.
- Prey L, Schmidhalter U. Simulation of satellite reflectance data using high-frequency ground based hyperspectral canopy measurements for in-season estimation of grain yield and grain nitrogen status in winter wheat. Isprs J Photogramm. 2019; 149: 176–187.
- Pullanagari RR, Kereszturi G, Yule IJ. Mapping of macro and micro nutrients of mixed pastures using airborne AisaFENIX hyperspectral imagery. Isprs J Photogramm. 2016; 117 (1–10)

- Rasmussen, CE, Williams, CKI. Gaussian Processes for Machine Learning. The MIT Press; USA: 2006.
- Rast M, Painter T. Earth observation imaging spectroscopy for terrestrial systems: an overview of its history, techniques, and applications of its missions. Surv Geophys. 2019; 40: 303–331.
- Richter K, Timmermans WJ. Physically based retrieval of crop characteristics for improved water use estimates. Hydrol Earth Syst Sci. 2009; 13: 663–674.
- Richter K, Hank TB, Mauser W, Atzberger C. Derivation of biophysical variables from earth observation data: validation and statistical measures. J Appl Remote Sens. 2012a; 6: 23.
- Richter K, Hank TB, Vuolo F, Mauser W, D'Urso G. Optimal exploitation of the sentinel-2 spectral capabilities for crop leaf area index mapping. Remote Sens. 2012b; 4: 561–582.
- Rivera-Caicedo JP, Verrelst J, Muñoz-Marí J, Camps-Valls G, Moreno J. Hyperspectral dimensionality reduction for biophysical variable statistical retrieval. Isprs J Photogramm. 2017; 132: 88–101.
- Rosati A, Day KR, DeJong TM. Distribution of leaf mass per unit area and leaf nitrogen concentration determine partitioning of leaf nitrogen within tree canopies. Tree Physiol. 2000; 20: 271–276. [PubMed: 12651464]
- Sample R, Babst BA. Timing of nitrogen resorption-related processes during fall senescence in southern oak species. For Sci. 2018; 65: 245–249.
- Schaepman, ME; Green, R; Ungar, S; Curtiss, B; Boardman, J; Plaza, A; Gao, B-C; Ustin, S; Kokaly, R; Miller, J; Jacquemoud, S; , et al. The future of imaging spectroscopy prospective technologies and applications; IEEE International Symposium on Geoscience and Remote Sensing (IGARSS); Denver, CO, USA. 2006. 2005–2009.
- Schnug E, Haneklaus S. Sulphur deficiency symptoms in oilseed rape (Brassica napus L.) the aesthetics of starvation. Phyton-Annales Rei Botanicae. 2005; 45: 79–95.
- Scott Green D, Erickson JE, Kruger EL. Foliar morphology and canopy nitrogen as predictors of light-use efficiency in terrestrial vegetation. Agric For Meteorol. 2003; 115: 163–171.
- Serrano L, Peñuelas J, Ustin SL. Remote sensing of nitrogen and lignin in Mediterranean vegetation from AVIRIS data: decomposing biochemical from structural signals. Remote Sens Environ. 2002; 81: 355–364.
- Skiba U, Rees B. Nitrous oxide, climate change and agriculture. CAB Review. 2014; 9: 1–7.
- Smola AJ, Schölkopf B. A tutorial on support vector regression. Stat Comput. 2004; 14: 199–222.
- Söderström M, Piikki K, Stenberg M, Stadig H, Martinsson J. Producing nitrogen (N) uptake maps in winter wheat by combining proximal crop measurements with Sentinel-2 and DMC satellite images in a decision support system for farmers. Acta Agriculturae Scandinavica, Section B— Soil & Plant Science. 2017; 67: 637–650.
- Somers B, Asner GP, Tits L, Coppin P. Endmember variability in spectral mixture analysis: a review. Remote Sens Environ. 2011; 115: 1603–1616.
- Song S, Gong W, Zhu B, Huang X. Wavelength selection and spectral discrimination for paddy rice, with laboratory measurements of hyperspectral leaf reflectance. Isprs J Photogramm. 2011; 66: 672–682.
- Stroppiana D, Boschetti M, Brivio PA, Bocchi S. Estimation of Plant Nitrogen Concentration in paddy rice from field canopy spectra. Ital J Remote Sens. 2009a; 41: 45–57.
- Stroppiana D, Boschetti M, Brivio PA, Bocchi S. Plant nitrogen concentration in paddy rice from field canopy hyperspectral radiometry. Field Crops Res. 2009b; 111: 119–129.
- Suits GH. The calculation of the directional reflectance of a vegetative canopy. Remote Sens Environ. 1972; 2: 117–125.
- Thenkabail P, Mariotto I, Gumma M, Middleton E, Landis D, Huemmrich K. Selection of Hyperspectral Narrowbands (HNBs) and composition of Hyperspectral Twoband Vegetation Indices (HVIs) for biophysical characterization and discrimination of crop types using field reflectance and Hyperion/EO-1 data. Selected Topics in Applied Earth Observations and Remote Sensing, IEEE Journal of. 2013a; 6: 427–439.
- Thenkabail PS, Mariotto I, Gumma MK, Middleton EM, Landis DR, Huemmrich KF. Selection of Hyperspectral Narrowbands (HNBs) and composition of Hyperspectral Twoband Vegetation Indices (HVIs) for biophysical characterization and discrimination of crop types using field

reflectance and Hyperion/EO-1 data. IEEE J Sel Top Appl Earth Obs Remote Sens. 2013b; 6: 427–439.

- Thomas JR, Oerther GF. Estimating nitrogen content of sweet pepper leaves by reflectance measurements1. Agron J. 1972; 64: 11–13.
- Thorp KR, Wang G, West AL, Moran MS, Bronson KF, White JW, Mon J. Estimating crop biophysical properties from remote sensing data by inverting linked radiative transfer and ecophysiological models. Remote Sens Environ. 2012; 124: 224–233.
- Tian Y, Yao X, Yang J, Cao W, Zhu Y. Extracting red edge position parameters from ground- and space-based hyperspectral data for estimation of canopy leaf nitrogen concentration in rice. Plant Production Science. 2011; 14: 270–281.
- Tilly N, Bareth G. Estimating nitrogen from structural crop traits at field scale a novel approach versus spectral vegetation indices. Remote Sens. 2019; 11: 2066.
- Tuia, D, Volpi, M, Verrelst, J, Camps-Valls, G. Advances in Kernel Machines for Image Classification and Biophysical Parameter Retrieval. 2018. 399–441.
- Ustin SL, Roberts DA, Gamon JA, Asner GP, Green RO. Using imaging spectroscopy to study ecosystem processes and properties. Bioscience. 2004; 54: 523–534.
- van der Tol C, Verhoef W, Timmermans J, Verhoef A, Su Z. An integrated model of soilcanopy spectral radiances, photosynthesis, fluorescence, temperature and energy balance. Biogeosciences. 2009; 6: 3109–3129.
- Verger A, Baret F, Camacho F. Optimal modalities for radiative transfer-neural network estimation of canopy biophysical characteristics: evaluation over an agricultural area with CHRIS/PROBA observations. Remote Sens Environ. 2011; 115: 415–426.
- Verhoef W. Light scattering by leaf layers with application to canopy reflectance modeling: the SAIL model. Remote Sens Environ. 1984; 16: 125–141.
- Verhoef W, Jia L, Xiao Q, Su Z. Unified optical-thermal four-stream radiative transfer theory for homogeneous vegetation canopies. IEEE Trans Geosci Remote Sens. 2007; 45: 1808–1822.
- Verrelst J, Muñoz J, Alonso L, Delegido J, Rivera JP, Camps-Valls G, Moreno J. Machine learning regression algorithms for biophysical parameter retrieval: opportunities for sentinel-2 and -3. Remote Sens. Environ. 2012; 118: 127–139.
- Verrelst J, Camps-Valls G, Muñoz-Marí J, Rivera JP, Veroustraete F, Clevers JGPW, Moreno J. Optical remote sensing and the retrieval of terrestrial vegetation bio-geophysical properties - a review. Isprs J Photogramm. 2015; 108: 273–290.
- Verrelst J, Malenovský Z, Van der Tol C, Camps-Valls G, Gastellu-Etchegorry J-P, Lewis P, North P, Moreno J. Quantifying vegetation biophysical variables from imaging spectroscopy data: a review on retrieval methods. Surv Geophys. 2019; 40: 589–629.
- Vuolo F, Atzberger C, Richter K, D'Urso G, Dash J. Retrieval of biophysical vegetation products from RapidEye imagery. ISPRS International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences. 2010. 281–286.
- Wang F, Huang J-f, Wang Y, Zhanyu L, Zhang F. Estimating nitrogen concentration in rape from hyperspectral data at canopy level using support vector machines. Precis Agric. 2012; 14: 172– 183.
- Wang FM, Huang JF, Wang Y, Liu ZY, Peng DL, Cao FF. Monitoring nitrogen concentration of oilseed rape from hyperspectral data using radial basis function. International Journal of Digital Earth. 2013; 6: 550–562.
- Wang Z, Skidmore AK, Darvishzadeh R, Heiden U, Heurich M, Wang T. Leaf nitrogen content indirectly estimated by leaf traits derived from the PROSPECT model. IEEE J Sel Top Appl Earth Obs Remote Sens. 2015a; 8: 3172–3182.
- Wang Z, Skidmore AK, Wang T, Darvishzadeh R, Hearne J. Applicability of the PROSPECT model for estimating protein and cellulose+lignin in fresh leaves. Remote Sens Environ. 2015b; 168: 205–218.
- Wang J, Shen C, Liu N, Jin X, Fan X, Dong C, Xu Y. Non-destructive evaluation of the leaf nitrogen concentration by in-field visible/near-infrared spectroscopy in pear orchards. Sensors (Basel). 2017; 17: 538.

- Wang Z, Skidmore AK, Darvishzadeh R, Wang T. Mapping forest canopy nitrogen content by inversion of coupled leaf-canopy radiative transfer models from airborne hyperspectral imagery. Agric For Meteorol. 2018; 253–254: 247–260.
- Waske B, Fauvel M, Benediktsson J, Chanussot J. Machine Learning Techniques in Remote Sensing Data Analysis. 2009.
- Weiss M, Baret F, Myneni RB, Pragnere A, Knyazikhin Y. Investigation of a model inversion technique to estimate canopy biophysical variables from spectral and directional reflectance data. Agronomie. 2000; 20: 3–22.
- Weiss M, Jacob F, Duveiller G. Remote sensing for agricultural applications: a meta-review. Remote Sens Environ. 2020; 236 111402
- Williams, P, Norris, K. Near-infrared technology in the agricultural and food industries. American Association of Cereal Chemists; 1987.
- Wold S, Sjöström M, Eriksson L. PLS-regression: a basic tool of chemometrics. Chemometrics Intellig Lab Syst. 2001; 58: 109–130.
- Wright IJ, Reich PB, Westoby M, Ackerly DD, Baruch Z, Bongers F, Cavender-Bares J, Chapin T, Cornelissen JHC, Diemer M, Flexas J, et al. The worldwide leaf economics spectrum. Nature. 2004; 428: 821–827. [PubMed: 15103368]
- Xu C, Fisher R, Wullschleger SD, Wilson CJ, Cai M, McDowell NG. Toward a mechanistic modeling of nitrogen limitation on vegetation dynamics. PLoS One. 2012; 7 e37914 [PubMed: 22649564]
- Yamashita T, Yamashita K, Kamimura R. A stepwise AIC method for variable selection in linear regression. Communications in Statistics—Theory and Methods. 2007; 36: 2395–2403.
- Yang G, Zhao C, Pu R, Feng H, Li Z, Li H, Sun C. Leaf nitrogen spectral reflectance model of winter wheat (Triticum aestivum) based on PROSPECT: simulation and inversion. J Appl Remote Sens. 2015; 9 095976
- Yao X, Ren H, Cao Z, Tian Y, Cao W, Zhu Y, Cheng T. Detecting leaf nitrogen content in wheat with canopy hyperspectrum under different soil backgrounds. Int J Appl Earth Obs Geoinf. 2014; 32: 114–124.
- Yao X, Huang Y, Shang G, Zhou C, Cheng T, Tian Y, Cao W, Zhu Y. Evaluation of six algorithms to monitor wheat leaf nitrogen concentration. Remote Sens. 2015; 7 14939
- Yasumura Y, Hikosaka K, Hirose T. Nitrogen resorption and protein degradation during leaf senescence in Chenopodium album grown in different light and nitrogen conditions. Funct Plant Biol. 2007; 34: 409–417. [PubMed: 32689368]
- Yeoh H-H, Wee Y-C. Leaf protein contents and nitrogen-to-protein conversion factors for 90 plant species. Food Chem. 1994; 49: 245–250.
- Yi Q, Huang J, Wang F, Wang X. Evaluating the performance of PC-ANN for the estimation of rice nitrogen concentration from canopy hyperspectral reflectance. Int J Remote Sens. 2010; 31: 931–940.
- Yu K, Li F, Gnyp ML, Miao Y, Bareth G, Chen X. Remotely detecting canopy nitrogen concentration and uptake of paddy rice in the Northeast China Plain. Isprs J Photogramm. 2013; 78: 102–115.
- Zheng H, Cheng T, Li D, Zhou X, Yao X, Tian Y, Cao W, Zhu Y. Evaluation of RGB, color-infrared and multispectral images acquired from unmanned aerial systems for the estimation of nitrogen accumulation in rice. Remote Sens. 2018; 10: 824.
- Zhou K, Cheng T, Zhu Y, Cao WX, Ustin SL, Zheng HB, Yao X, Tian YC. Assessing the impact of spatial resolution on the estimation of leaf nitrogen concentration over the full season of paddy rice using near-surface imaging spectroscopy data. Front Plant Sci. 2018; 9



Fig. 1. Nitrogen as key component of biomass (left) and its proportional allocation in a C3 plant leaf (right). Adapted from Chapin et al. (1987).



Fig. 2.

Relationship between field measured canopy chlorophyll content (i.e. leaf area index, LAI, multiplied with leaf chlorophyll content, C_{ab}) and measured aboveground N content (N_{area}) as sum of measured N content of leaves, stalks and ears. Regression line in green shows only dates with solely green vegetation (vegetative stage). Regression line in black was achieved using all dates of the MNI field campaign (winter wheat from 29/3–6/7 2017 and maize from 17/7–15/9 2017), as described in Berger et al. (2018b) (Unpublished data). (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

Berger et al.



Fig. 3. Absolute number of N retrieval studies per year from 1994 to 2018 as identified by the literature review.

Berger et al.



Fig. 4.

Sensor systems used in hyperspectral N retrieval studies. Field: field-based spectrometers, Lab: laboratory analysis (with spectrometers), UAV: sensors on unmanned aerial vehicles, Airborne: sensors on aircrafts, Satellite: satellite-based sensors, Synthetic: simulated reflectance data from hyperspectral sensors, e.g. by RTMs.

Berger et al.



Fig. 5. Identified methods used for the estimation of N content and N concentration.

Page 40 Parametric regressions Total number of studies Chemometrics Nonparametric nonlinear regressions RTMs Hybrid or combined methods

Fig. 6. Development of the five main categories of N retrieval methods in scientific studies over time (from 1994 to 2018).

Berger et al.



Fig. 7.

Total number of studies apparently retrieving N content, N concentration and N uptake. The color of the bars indicates the in reality estimated measure. For instance, the large part in darker green of the first column (N content) indicates all those studies which in reality estimated N% despite the use of "N content" as terminology. In some cases, more than one term was used in a single study. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

Table 1

Specific absorption bands associated with proteins and nitrogen and the related absorption mechanisms according to the literature (Curran, 1989; Fourty et al., 1996; Kumar et al., 2001). Wavelength in brackets corresponds to the values remeasured by Fourty et al. (1996).

Wavelength (nm)	Absorption mechanism/electron transition	Absorbing compound		
910	C—H stretch, 3rd overtone	Protein		
1020	N—H stretch	Protein		
1510 (1520)	N—H stretch, 1st overtone	Protein, nitrogen		
1690	C—H stretch, 1st overtone	Protein, nitrogen, lignin, starch		
1730	C—H stretch	Protein		
1940	O—H stretch, O—H deformation	Protein, nitrogen, water, lignin, cellulose, starch		
1980 (1960)	N—H asymmetry	Protein		
2060	N—H stretch, N=H rotation	Protein, nitrogen		
2130	N—H stretch	Protein		
2180 (2200)	N—H rotation, C—H stretch, C—O stretch, C=O stretch	Protein, nitrogen		
2240 (2270)	C—H stretch	Protein		
2300 (2290)	C-H rotation, C=O stretch, N-H stretch	Protein, nitrogen		
2350	CH2 rotation, C-H deformation	Protein, nitrogen, cellulose		

Table 2

Overview of main retrieval methods used for N sensing from hyperspectral data in an agricultural context. Categorization according to Verrelst et al. (2019).

Category	Method group, abbreviation	Exemplary references		
Linear parametric regressions	Narrowband reflectance or derivative types	(Alchanatis et al., 2005; Goel et al., 2003; Lee et al., 2008)		
	Narrowband ratio vegetation indices, NB-VI	(Chen et al., 2010; Herrmann et al., 2010)		
	Spectral shapes/transformation	(Feng et al., 2014; Yao et al., 2015)		
Linear nonparametric regressions (chemometrics)	(Stepwise) multiple linear regression, (S)MLR	(Castaldi et al., 2016; Miphokasap and Wannasiri, 2018)		
	Principal component analysis/regression, PCA/PCR	(Wang et al., 2017)		
	Partial least squares regression, PLSR	(Hansen and Schjoerring, 2003; Inoue et al., 2012)		
Nonlinear nonparametric regressions (machine learning)	Decision trees (e.g. Random forest regression, RFR)	(Näsi et al., 2018; Pullanagari et al., 2016)		
	Artificial neural networks, ANN	(Afandi et al., 2016; Monteiro et al., 2007)		
	Kernel-based (Support vector regression, SVR; Gaussian processes regression, GPR)	(L. Li et al., 2018; Wang et al., 2012; Zhou et al., 2018)		
Physically-based methods	Radiative transfer modelling, RTM	(Z. Li et al., 2018; Yang et al., 2015)		
Hybrid methods	Combination of at least two methods in synergic use	(Song et al., 2011; Thorp et al., 2012)		

Table 3

Overview of recently launched or planned spaceborne imaging spectrometer missions which will potentially provide opportunities for complementary global N sensing. SSD: spectral sampling distance.

Mission (organisation, country)	Spectral range, (SSD, no. of bands)	Spatial resolution (swath)	Launch date	Usage	Reference
PRISMA (ASI, Italy)	400–2500 nm (6–12 nm, 237 bands)	30 m (30 km)	22/03/2019	Technology demonstrator	(Loizzo et al., 2019)
HISUI (METI, Japan),	400–2500 nm (10–12 nm, 185 bands)	20 m (cross-track) × 30 m (along-track) (20 km)	05/12/2019	Operational	(Matsunaga et al., 2017)
EnMAP (DLR, Germany)	400–2500 nm (6.5–10 nm, 242 bands)	30 m (30 km)	04/2021	Scientific precursor	(Guanter et al., 2015)
SHALOM (Italian- Israeli)	400–2500 nm (10 nm, 275 bands)	10 m (30 km)	2022	Operational/ commercial	(Feingersh and Ben Dor, 2015)
CHIME (ESA)	400-2500 nm (210 bands)	20–30 m	tbd	Copernicus high- priority mission candidate	(Nieke and Rast, 2018)
SBG (NASA, U.S.)	350/400–2500 nm (10 nm or better, 210 bands)	30–45 m (150 km)	tbd	Operational	(NASA, 2018)